$(ov\sigma)$ 

Social Survey Division

# Family Expenditure Survey 2000-2001

VOLUME 5
Database Changes

# **DATABASE CHANGES**

Contents	Colour
Explanatory note	pink
Raw variable changes	green card
Part 1 - New raw variables for 2000-2001	yellow
Part 2 - 1999-2000 raw variables deleted for 2000-2001	pink
Part 3 - 1999-2000 raw variables amended for 2000-2001	light green
Derived variable changes	orange card
Part 4 - New derived variables for 2000-2001	yellow
Part 5 - 1999-2000 derived variables deleted for 2000-2001	pink
Part 6 - 1999-2000 derived variables amended for 2000-2001	green
Expenditure code changes	blue card
Part 7 - New expenditure codes for 2000-2001	yellow
Part 8 - 1999-2000 expenditure codes deleted for 2000-2001	pink
Part 9 - 1999-2000 expenditure codes amended for 2000-2001	green
Other changes	white card
Part 10 - New tables for 2000-2001	yellow
Part 11 - 1999-2000 tables deleted for 2000-2001	pink
Part 12 - Major changes (1991 to 2000-2001)	green
Part 13 - Questionnaire changes for 2000-2001	blue

#### **DATABASE CHANGES**

#### **Explanatory Notes**

This volume of the documentation describes the way in which the 2000-2001 database has changed in relation to the 1999-2000 database. Changes are described in terms of the names of the affected variables and codes. This volume is divided into 13 parts.

- Part 1 Changes to raw variables are listed in terms of those that are new for 2000-2001.
- Part 2 Lists the variables that have been deleted from 1999-2000
- Part 3 Lists any variables that have been amended from 1999-2000, including a description of the change.

#### Parts 4-6 - Changes to derived variables

Many derived variables remain unchanged in name, but it should be remembered that there may well be changes in the contents of the variables that feed into them. You should check the compositions of the variables you intend using. Changes to derived variables are listed that:

- Part 4 are new for 2000-2001
- Part 5 have been deleted from 1999-2000 and
- Part 6 which have been amended from 1999-2000, including a description of the change.

#### Parts 7-9 - FES expenditure codes

Expenditure codes are listed and explained in Volume 4 of the User Documentation.

- Part 7 details expenditure codes which are new to 2000-2001
- Part 8 details expenditure codes which have been deleted from 1999-2000 and
- Part 9 shows expenditure codes which are amended from 1999-2000, with details of the changes.

#### Parts 10-11 - Changes to tables

- Part 10 details new tables which have been set up for 2000-2001 and describes their contents.
- Part 11 details tables that have been deleted from 1999-2000.

Date Produced: 21 September 2001

#### **DATABASE CHANGES**

## Parts 12-13 - Major changes and changes to the Questionnaires

- **Part 12** details of the changes to the Family Expenditure Survey over a ten year view from 1991 to the current year 2000-2001.
- **Part 13** details the changes to the questionnaires in 2000-2001 and is based upon a paper written by the Social Survey Division.

Date Produced: 21 September 2001

Variable	Description	Table Name	Question Reference
Household Quest	ionnaire		
access0	How does your household access the inter	hhold2	Н 190.80
access1	How does your household access the inter	hhold2	H 190.80
access2	How does your household access the inter	hhold2	Н 190.80
access3	How does your household access the inter	hhold2	Н 190.80
access4	How does your household access the inter	hhold2	H 190.80
ccint_anymore	Have you paid any more of these card acc	ccint	H 340.65
ccint_cardnum	Card Number	ccint	H 340.45
cdplay	Compact disc (CD) player?	hhold2	Н 190.65
hihnum	You have told me thatjointly own or	narcon	H 20.51
mmum	Tou have told the thatjointly own of	person	11 20.31
interest	Was any INTEREST shown on the account(s)	hhold2	H 340.21
intamt	How much interest was shown?	ccint	Н 340.55
jntelda	Enter person number of the eldest joint	person	H 20.52
jnteldb	Enter person number of the eldest joint	person	H 20.53
numhhdef	How many groups are there within this	person	H 20.38
oldhhdef	Do all household members share the living	person	Н 20.37
paycard	Have (any of) you PAID a credit charge	hhold2	Н 340.16
satcab0	"TV satellite/cable, digital receiver "	hhold2	H 190.15
satcab1	"TV satellite/cable, digital receiver "	hhold2	H 190.15
satcab2	"TV satellite/cable, digital receiver "	hhold2	H 190.15
satcab3	"TV satellite/cable, digital receiver "	hhold2	H 190.15
wwweb	Are you able to access the web from home	hhold2	Н 190.85
Income Question	naire		
anymore	Do you have any more incomes?	minine	I 50.460
-	•		7.000.10
cashamt	How much sent to charity abroad	variousi	I 280.10
cctcrdt	Do either of these payments include a Chrg	paymain	I 50.288
dedtot	How much was deducted for charities	paymain	I 50.80
dptcamt	How much was included for DPTC payment?	paymain	I 50.285
dptcper	What period did this cover?	paymain	I 50.286
dvdptc	WE of Disable person's tax credit	paymain	I 50.287
dvdptc2	WE of Disabled person's tax credit	benr1	I 140.95
dvwftc	WE of Working Family tax credit	paymain	I 50.284
grosamt	What was the gross amount paid?	mininc	I 50.440
grosper	What period did this cover	mininc	I 50.445
incnum	Income number	mininc	I 50.431
mininc_source	Income Source	minine	I 50.435

Variable	Description	Table Name	Question Reference
netamt	What was the net amount paid?	mininc mininc	I 50.450
netper	What period did this cover?	mininc	I 50.455
payerdt	Did the take home pay of YPayAmt include	paymain	I 50.281
person	Person number (ie) who receives income?	mininc	I 50.430
wftcamt	How much was included for WFTC payment?	paymain	I 50.282
wftcper	What period did this cover ?	paymain	I 50.283
Diary Questionnai	re		
busrfnd_day	Day number of Business Refund	busrfnd	D 60.5
busrfnd_itemnum	Business refund item number	busrfnd	D 60.0
eocode	Expenditure Code	eout	D 50.35
eout_day	"If entry for a day, key the day number"	eout	D 50.5
eout_foodcod	Food Code	eout	Not used
eout_foodtyp	Enter the expenditure description	eout	D 50.10
eout_hotcold	Was the item (or most of it) hot or cold?	eout	D 50.25
eout_itemnum	Item Number	eout	D 50.0
onoff	Consumed on or off the premises?	eout	D 50.20
outlet	Where was the item bought?	eout	D 50.15
payeout	What is the total amount paid?	eout	D 50.40
payyouth	How much did you pay?	youth	D 55.25
refcode	Expenditure code	busrfnd	D 60.12
refdes	Describe what the business refund is for	busrfnd	D 60.10
refndamt	How much was refunded?	busrfnd	D 60.20
refnet	Calculation for net expenditure	busrfnd	D 60.25
refpaid	How much did you pay ?	busrfnd	D 60.15
typealc	What type of alcohol was consumed away from home	eout	D 50.30
youth_day	YOUTH DIARY: EATING OUT SPEND	youth	D 55.5
youth_foodcod	Food Code (expenditure code)	youth	D 55.20
youth_foodtyp	Enter the expenditure description	youth	D 55.10
youth_hotcold	Was the item (or most of it) hot or cold	youth	D 55.15
youth_itemnum	Youth Item number	youth	D 55.0

## 1999-2000 Raw variables that have been deleted in 2000-2001

Variable	Description	Table Name
		.,
allow_perwho	Which person gave you the allowance?	allow
allwfor	Allowance: Who are the payments for?	sep
anyint	If interest paid on credit cards,Code1	intcred
anywin	If winnings in last 14 days code 1	winnings
bkpers01	Bank charges person number	bk
bkpers02	Bank charges person number	bk
bkpers03	Bank charges person number	bk
bkpers04	Bank charges person number	bk
bkpers05	Bank charges person number	bk
bkpers06	Bank charges person number	bk
bkpers07	Bank charges person number	bk
bkpers08	Bank charges person number	bk
bkpers09	Bank charges person number	bk
bkpers10	Bank charges person number	bk
bkpers11	Bank charges person number	bk
bldamt	What interest recvd from Bld Soc accnts?	variousi
bldjnt	B/Soc accnt: held solely or jointly?	variousi
bldtax	B/Soc accnt: is int paid pre/post tax?	variousi
borramt	Amnt originally borrowed to buy house	mort
bothac	Any interest from other bank / society?	variousi
bothamt	What interest recvd from other banks etc	variousi
bothnt	Other bank accs: held solely or jointly?	variousi
buildac	Any interest credited to B/Soc account?	variousi
buytype	Did you originally buy this house/flat?	mort
buyyear	In which year did you buy house/flat?	mort
cdplayer	Do any of you have a Compact Disc player	hhold2
chtype	What fuel does the central heating use?	chi
chyamt	How much have you sent abroad to charity	variousi
clubtype	Which of these arrangements was used	lastmth
credit	Was item bought with a credit/charge crd	expend
curroth	What is the name of the Bank/Bldg Socy?	bk
currtype	Is your account with a Bank or bldg soc?	bk
dchftsam	How much was deducted fr charities gross	paymain
dchocsam	How much was deducted fr other charities	paymain
dedchocs	Any deductions for other charities?	paymain
dedchtfs	Any tax free deductions for charities?	paymain
disposal	Does child usually spend or save monies?	childinc
dvbaby	Baby sitting, mail order in last 12 months	variousi
dvbld1	Building Society interest rcvd, last 12m	variousi
dvchft	WEV of deductions for charities.	paymain
dvcho	WEV for deductions for other charities.	paymain
dvchy	Amount sent abroad to charity, last 12m	variousi
dvdw	Disab. working allowance amount received	benr1
dvechg	WE of electricity charge, BBS/COCD N.Ire	hhold2
dvgch	WE of gas charge, Board Budgeting Scheme	hhold2
dvhbso	WEV of amnt deducted for other services	rents1
dvhola	WE of package holiday, full adult rate	holpack
dvholc	WE of package holiday, reduced rate	holpack
dviwc	WE of amount to claim for dec, repairs	dec
dvmot	WEV of motoring expenses in usual pay	paymain
dvoth1	Other bank/society interest, last 12 mth	variousi
dvreg	WEV of regular paid work.	ilo
dvretn	Amount earned in 12m prior to retiring	ilo
	Γ	

## 1999-2000 Raw variables that have been deleted in 2000-2001

Variable	Description	
lvrtws	Rent: Amnt incl for water/sewerage.	rents1
lvtelc	WE of telephone charge, Budgeting Scheme	phone
lvtrav	WEV of cost of school travel	trav
vuav	WEV of cost of school daver	uuv
chgamt	What was the charge on yr lst electric?	hhold2
chgper	What period did this cover?	hhold2
mpfreem	How many free meals have you received?	paymain
ndwprin	How is repayment of orig loan covered?	mort
xrent	Were you renting this hse before buying?	mort
ridge	Do you have a refridgerator in yr hhold?	hhold2
chgamt	How much was the charge on yr lst advice	hhold2
chgper	What period did this cover?	hhold2
wkinc	Computed gross weekly household income	impute
nbsoamt	Total amount deducted for this service?	rents1
bsother	What is/are this/these service(s)?	rents1
nbsvoth	Any other services included in the rent?	rents1
hedfilt	Were any values imputed by the editor?	impute
hold_filter13	Childrens travel to/from school	hhold2
oladram	Amount of holiday payment at adult rate	holpack
nolchr	Number of reduced rate persons in pmnt	holpack
nolchram	Amount of holiday pmnt at reduced rate	holpack
nolstrtm	Month holiday start(s)?	holpack
olstrty	Year holiday started	holpack
na207	What is the reason for your absence?	ilo
na209	Are you receiving any pay?	ilo
na217	Unable to work, due to sick relative?	ilo
na275	Have you done this work in last 7 days?	variousi
nco	Describe other sources of income.	sejob
ndper	Who gave you this payment?	allow
nswille	DECS: Will you be claiming on insurance?	dec
ntcod	IntCod	intcred
ntcred_itemnum	Interest on credit cards item number	intcred
ntl12m	How much interest paid on the mortgage?	mort
ntmth ntnorbm	How many months does this cover?	mort
ntperbm ntperby	When did that 12 month period begin? When did that 12 month period begin?	mort
ntperby nvflt	TESSA: Amount invested in last 12 months	mort variousi
temdea_lns	Items bt with a loan - was it new/2nd?	itemdea
temdea_ms temdea_supplier	Item loan: type of firm item sourced?	itemdea
wcamt	DECS: How much will you be claiming?	dec
		: 10
ob2_director	In this job are you a director?	job2
ob2_dirni	Were NI deductions deducted at source?	job2
ob2_indd	What did the firm/org you worked for do?	job2
ob2_indt	Short title for the industry	job2
ob2_occd	What did you mainly do in your sec. job?	job2
ob2_sect1	What was the type of org. you worked in?	job2
ob2_sect2	What type of non-private organisation?	job2
ob2_workstat ob3_director	Were you an employee or self-employed?	job2
_	In this job are you a director?  In this job were NI contribs ded at srce	job3 job3
ob3_dirni		

## 1999-2000 Raw variables that have been deleted in 2000-2001

Variable	e Description	
job3_indt	Short title for the industry	job3
job3_occd	What did you mainly do in your third job	job3
job3_sect1	What type of organisation do you wrk for	job3
job3_sect2	What type of non-private organisation?	job3
ob3_workstat	Are you an employee or self-employed?	job3
obmain_sect1	What type of organisation do you work fr	jobmain
obmain_sect2	What kind of organisation was it?	jobmain
astamt	How much was the most recent instalment?	lastpay
ender	Who currently provides the mortgage/loan	mort
nailbab1	Coded as mail order if working as both	variousi
nailbaby	Rcvd monies from baby sitting/mail order	variousi
nkrc127	IntL12M has been abated)	mort
nkrc191a	GChgAmt has been abated)	hhold2
nkrc201a	EChgAmt has been abated)	hhold2
nkrc264	Abatement marker for phone charge	phone
nkrhp1	Part exchange allowance has been abated	agree1
nlbabamt	How much in total earned in last 12 mths	variousi
nntwho	Who is the maintenance for?	allow
nobile	Do you have a mobile phone in yr hhold?	hhold2
nort_contrb10	How much does (pers) contrib to mortgage	mort
nort_contrib	How much does (pers) contrib to mortgage	mort
nort_contrib1	How much does (pers) contrib to mortgage	mort
nort_contrib2	How much does (pers) contrib to mortgage	mort
nort_contrib3	How much does (pers) contrib to mortgage	mort
nort_contrib4	How much does (pers) contrib to mortgage	mort
_		
nort_contrib5	How much does (pers) contrib to mortgage	mort
nort_contrib6	How much does (pers) contrib to mortgage	mort
nort_contrib7	How much does (pers) contrib to mortgage	mort
nort_contrib8	How much does (pers) contrib to mortgage	mort
nort_contrib9	How much does (pers) contrib to mortgage	mort
nortcont	Anyone other than HOH pay mortgage?	mort
nortlen	Year mortgage began	mort
nortnam1	Who contributes towards the mortgage?	mort
mortnam2	Who contributes towards the mortgage?	mort
nortnam3	Who contributes towards the mortgage?	mort
nortnam4	Who contributes towards the mortgage?	mort
nortnam5	Who contributes towards the mortgage?	mort
nortnam6	Who contributes towards the mortgage?	mort
nortnam7	Who contributes towards the mortgage?	mort
nortnam8	Who contributes towards the mortgage?	mort
nortnam9	Who contributes towards the mortgage?	mort
nortname	Who contributes towards the mortgage?	mort
nortnm10	Who contributes towards the mortgage?	mort
not	MOT included in vehicle service	serv
noteamt	How much was incd for motoring expenses	paymain
ewo	New or second hand item or service obtained	agree1
iewold	New or second hand item or service obtained	agree2
ohhatad	How many other households live here?	address
noinc0	No income from self-employment.	sejob
noinc1	No income from self-employment.	sejob
noinc2	No income from self-employment	sejob
noinc3	No income from S-E - other sources?	sejob

Date Amended: 21 September 2001 Page 3

# 1999-2000 Raw variables that have been deleted in 2000-2001

Variable	Description	Table Name	
otherhed	Which other questions imputed/amended?	impute	
thholiv	Hholds besides your own living here?	address	
othsrv	What is/are this/these other services?	rents1	
ayint	Amount of interest paid on credit card	intcred	
aymain_freemeal	Have you received any free meals?	paymain	
aymeth	Payment method for vehicle purchase	vbought	
paywin	Amount of winnings	winnings	
ensdatm	Month private personal pension started	pendet	
ensdaty	Year private personal pension started	pendet	
erc400	How often were you usually paid?	ilo "	
oolco	Name of insurance company	medins	
rivamt	Amount sent abroad to a pvt individual	variousi	
orot0001	Mort. prot. policy: type of policy	mortprot	
orot0002	Mort. prot. policy: type of policy	mortprot	
rottype	Mort. prot. policy: type of policy	mortprot	
texhp	How much allowed in part exchange	agree1	
urcamt	What was the purchase price of the house	mort	
ualif	Key qualifier code	expend	
edtimm	How many mnths were you working for firm	variousi	
edtimy	How many years were you working for firm	variousi	
efnd1	Key amount refunded or claimed for busin	expend	
egpay	What wage salary did you usually receive	ilo	
egwks	Number of weeks in regular paid work	ilo	
entfrom	Who was it rented from ?	mort	
t12m	Did you retire within last 12 months?	ilo	
tearn	How much did you earn prior to retiring?	ilo	
ntdish	Do you have a satellite TV dish?	hhold2	
erinc00	Card C1: Heating	rents1	
erinc01	Card C1: Lighting	rents1	
erinc02	Card C1: Hot water	rents1	
erinc03	Card C1: Fuel for cooking	rents1	
erinc04	Card C1: Food including prepared meals	rents1	
erinc05	Card C1: TV/Video rental	rents1	
erinc06	Card C1: TV license fees	rents1	
erinc07	Card C1: Personal laundry	rents1	
erinc08	Card C1: Medical expenses	rents1	
erinc09	Card C1: Nursing and personal care	rents1	
erinc10	Card C1: Transport	rents1	
erinc11	Card C1: Contents insurance	rents1	
erinc12	Card C1: Other services	rents1	
ervnum	Rents\\Services Card C1 Item Number	hbservam	
ublet	Have formal agreement to let/sublet?	address	
ıblety	To whom is the accommodation sublet?	address	
xrelal	Allowances for tax relief.	paymain	
xrelf	Standard tax relief ded from mort pymt?	mort	
elchamt	Phone: Amount of charge on last advice	phone	
elchper	Phone charge: what period was covered?	phone	
esdepos	TESSA: Was amnt a lump sum or instalmnts	variousi	
esinvst	TESSA: How much invested in last 12 mths	variousi	
essband	TESSA: In which group is your investmnt?	variousi	
av_anymore	Have any of you paid for school travel?	trav	

## 1999-2000 Raw variables that have been deleted in 2000-2001

Variable	Description	<b>Table Name</b>	
rav_num	Travel to school item number	trav	
ravamt	Trav: How much paid for (x) in lst 7 dys	trav	
ravdays	Trav: How many days did this cover?	trav	
ypecrse	Type of establishmnt where course ws run	edf2	
ypef	Firm type from which item obtained	agree1	
ypefirm	Firm type from which item or service obtained	agree2	
bought_vehtypeo	Other vehicle type bought in 1st 12 mths	vbought	
owned_vehtypeo	Other vehicle type owned last 12 months	vowned	
sold_vehtypeo	Other vehicle type sold in lst 12 months	vsold	
used_vehtypeo	Other vehicle type in continuous use	vused	
hathome	Type of dwelling in property fee	hexp2	
hichh00	RENT question imputed or amended (code)	impute	
hichh01	Housing benefit question imputed or amnd	impute	
hichh02	Services question imputed or amended	impute	
rhichh03	Wat/sew rates question imputed or amendd	impute	
hichh04	Price of house question imputed or amndd	impute	
hichh05	Mortgage pymnt question imputed or amndd	impute	
hichh06	Mortgage protectn question imputed/amndd	impute	
hichh07	Insurances: question imputed / amended	impute	
hichh08	CT band: question imputed or amended	impute	
hichh09	CT amount: question imputed or amended	impute	
hichh10	CT rebate: question imputed or amended	impute	
vhichh11 vhichh12	Life insurance: question imputed/amended	impute	
hichh13	Amnt of Life Ins: question imputed/amndd	impute	
hichh14	Last phone pymnt: question imputed/amndd	impute	
hichh15	Last electric pymnt: question imp/amndd Electricity chrg: question imptd/amndd	impute	
hichh16	Last gas pymnt: question imputed/amended	impute impute	
hichh17	Gas charge: question imputed/amended	impute	
hichh18	CH oil amount: question imputed/amended	impute	
hichh19	TV rental pymnt:question imputed/amended	impute	
hichh20	Road tax: question imputed/amended	impute	
hichh21	Vehicle ins: question imputed/amended	impute	
hichh22	Loan instalmnt: question imputed/amended	impute	
hichh23	Tax deducted: question imputed/amended	impute	
hichh24	NI deducted: question imputed/amended	impute	
hichh25	Superannuation: question imputed/amended	impute	
hichh26	Union fees: question imputed/amended	impute	
hichh27	Self employed tax: question imputed/amnd	impute	
hichh28	Self-emplyed NI: question imputed/amendd	impute	
hichh29	Retirement pension: question imptd/amndd	impute	
hichh30	Severe disab: question imputed/amended	impute	
hichh31	Disab Work All: question imputed/amended	impute	
hichh32	Disab Liv All: question imputed/amended	impute	
hichh33	Attend Allow: question imputed/amended	impute	
hichh34	JSA: question imputed/amended	impute	
hichh35	Income support: question imputed/amended	impute	
hichh36	Family credit: question imputed/amended	impute	
hichh37	Incapacity Ben: question imputed/amended	impute	
hichh38	Maternity allow:question imputed/amended	impute	
hichh39	Other imputed questions imputed or amndd	impute	
hopay	Who pays?	mort	
horecaa	Who receives the attendance allowance?	benr2	

Date Amended : 21 September 2001 Page 5

# 2000-2001 : SECTION 5, PART 2

## **Family Expenditure Survey**

## 1999-2000 Raw variables that have been deleted in 2000-2001

Variable	Description	Table Name	
whorecmc	For whom is the allowance received?	benr2	
whorecsc	For whom was the allowance received?	benr2	
whynopro	Why was respo unable to give a prft fig.	sejob	
wincod	WinCod	winnings	
winnings_itemnum	Winnings item number	winnings	
wintyp	Enter winnings code	winnings	
wksaw	Number of weeks away from work?	ilo	
worsinc	Were water/sewerage chrgs incd in rent?	rents1	
wsinc	Were water/sewerage charges inc in rent?	rents1	
wsincamt	How much was included for water/sewer?	rents1	

Page 6

Date Amended: 21 September 2001

Variable	Description	Table Name	<b>Description of Change</b>
Banding entry ch	anges		
incmstyr	Year mortgage protection policy started	mortprot	New banding in 2000-2001
linsstyr	Year policy taken out	life	New banding in 2000-2001
menstyr	Year policy taken out	endowp	New banding in 2000-2001
Changes to data t	ype		
incmstyr	Year mortgage protection policy started	mortprot	was integer2, is now integer1
linsstyr	Year policy taken out	life	was integer2, is now integer1
menstyr	Year policy taken out	endowp	was integer2, is now integer1
unearni1	Card U2 - any income as a sleeping partner	variousi	was integer, is now integer1
Variables with a	banding entry but no indication of a coding fram	me	
None.			
Variables where t	the banding entry is a coding frame in one year	but not the	other.
incmstyr	Year mortgage protection policy started	mortprot	New bandtype in 2000-2001
linsstyr	Year policy taken out	life	New bandtype in 2000-2001
menstyr	Year policy taken out	endowp	New bandtype in 2000-2001
Variables that ha	ve moved from one table to another.		
dvint	WEV of last 12 months interest paid	mort	1999-2000
dvint	WEV of last 12 months interest paid	ccint	2000-2001
Griff		Com	2000 2001

# **Family Expenditure Survey**

New derived variables

Variable	Description		Table Name
a003	Household reference person	n	set41
a0062	Ethnic origin of HoH and s	pouse	set41
a101	Telephone and\\or mobile i	n household	set1
a1642	Cable receiver in househole	d	set1
a1643	Digital receiver in househo	ld	set1
a190 a191 a192 a193 a194 a195	Internet access via Home C Internet access via Digital Internet access via mobile Internet access via games of Internet access via other modernet access via home co	TV phone console ethod	set1 set1 set1 set1 set1 set1
a2615	Bank\\Build Socs Acc with	int before tax	set46
a2616	Bank\\Build Soc Acc with	int after tax	set46
b237	Credit card interest paymen	nts	set24
b371	WFTC - last amount receiv	red in pay	set44
b372	DPTC - last amount receive	ed in pay	set44
b413	Int rcvd - Bank\\Build Soci	ety before tax	set46
b414	Int rcvd - Bank\\Build Soci	ety after tax	set46
d110210t	School Travel		set14t
fs1101 fs1301 fs1302 fs1303 fs1304 fs146	Gross Rent Mortage interest payments Net Rent Council tax, domestic rates Water & sewerage charges Miscellaneous child exp WEV of deductions for cha	[Family Spending only [Family Spending only [Family Spending only [Family Spending only	y] set26a y] set26a y] set26a y] set26a y] set26a
wevchty	WEV of deductions for cha	arities	setwrk

Date Produced: 24 August 2001

## 1999-2000 derived variables that have been deleted for 2000-2001

Variable	Description	Table Name	
a102	Talanhana in hausahald	cot1	
a119	Telephone in household Mortgage - source	set1 set12	
a1331	mortgage - source mortgage - year current mortgage began	set12	
a163	Tax relief deducted from mortgage?	set12	
a165	Fridge in household	set1	
a1651	Mobile phone in household (exc. car phon	set1	
a181	Year accommodation bought	set12	
a188	Owned outright - how acquired	set12	
a207	Reason away from work	set12 set43	
a209	Pay received while away from work - type	set43	
a212	Subsidiary occupation as an employee	set45	
a214	Subsidiary occupation as self employed	set45	
a2606		set46	
a2610	Interest credited to other savings acc	set46	
a2611	Building society account before tax Building society account after tax	set46	
	· ·		
a262	TESSA how amount was invested	set46	
a275	Mailorder agent\\babysitter - at present	set41	
a280	Maintenance separation allowance	set46	
a297	Maintenance expenditure - who paid to	set54	
abil	Other ins. premiums exc. Pers. Accident.	setabi	
abi2	Personal Accident policies.	setabi	
b111	Repairs-main dwell amnt reclaimed ins co	set12	
b112	Repairs-scnddwell amnt reclaimed ins co	set12	
b131	Amount originally borrowed	set12	
b132	Purchase price of dwelling	set12	
b140	Mortgage tax relief (non-miras only)	set12	
b141	Endow mortgage: net int (after tax rel)	set12	
b142	Repay mortgage: net int (after tax rel)	set12	
b158	School travel - amount paid last week	set14	
b158c	School travel - amount paid last week	set14c	
b158t	School travel - amount paid last week	set14t	
b167	Telephone budget a\\c-charge (ex mobiles)	set15	
b1671	Mobile phone budget a/c charge	set15	
b180	Bank charges - net amount last 3 months	set15	
b1801	Building society charges - net lst 3 mth	set15	
b223	Gas-amnt charged last board bdgt scheme	set13	
b224	Elec-amnt charged last board bdgt scheme	set13	
b272	Legal fees - main dwelling	set15	
b274	Legal fees-second dwelling	set15	
b327	Self-supply goods-value from d books	set43	
b345	Non-tax deductions from pension	set47	
b374	Other savings bank\\soc-interest recvd	set46	
b383	Mailorder agent\\babysitter amnt recvd	set46	
b395	TESSA amount invested in last year	set46	
b397	Childrens income - amount spent	set53	
b397t	Childrens income - amount spent	set53t	
b397th	Children income - amt spent hhold	set26a	
b399	Building soc before tax - interest rec	set46	
b400	Building soc after tax - interest rec	set46	
b401	High Street Bank - interest received before tax	set46	
b402	High Street Bank - interest received after tax	set46	
crtranty	Instalement credit\club - club type acquired from	set87	

Date amended: 23 August 2001 Page 1

## 1999-2000 derived variables that have been deleted in 2000-2001

Variable	Description	Table Name	
1140402t	Credit card interest payment	set24t	
11404021	Credit card interest payment	501241	
s147	Miscellaneous child expenditure	set26a	
s15621	Building Society	set26a	
s15622	Bank	set26a	
s15623	Other lenders	set26a	
054	Subsidiary job held	set41	
080	Building society acnts - interest & tax	set46	
102	Housing - credit expenditure	set54	
103	Fuel etc - credit expenditure	set54	
104	Food - credit expenditure	set54	
105	Alcoholic drink - credit expenditure	set54	
106	Tobacco - credit expenditure	set54	
107	Clothing & footwear - credit expenditure	set54	
112	Miscellaneous - credit expenditure	set54	
115	Second dwelling - credit expenditure	set54	
145	Household goods - credit expenditure	set54	
146	Household services - credit expenditure	set54	
147	Personal goods & services - credit expenditure	set54	
148	Motoring expenditure - credit expenditure	set54	
149	Fares & other - credit expenditure	set54	
150	Leisure goods - credit expenditure	set54	
151	Leisure services - credit expenditure	set54	
152	Credit component of total personal expendture	set54	
169	Current earnings-mailorder\\baby sitter	set46	
175	Social security benefits	set51	
316	<del>-</del>	set3	
	Imputed income from schedule b - hhld Imputed income from schedule b - head	set8	
317 318	Imputed income from schedule b - wife	set10	
	<u> •</u>	set11	
319	Imputed income from schedule b - others	set1	
395 514	Disposable income+self supply & in kind	set12	
514	Mortgage intrst\\prncpl - last instlmnt	Set12	
01	Cars, vans and motorcycle purchases	set26	
.011	New cars and vans	set26	
1012	Second hand cars and vans	set26	
1013	Motorcycles and scooters	set26	
102	Spares and accessories	set26	
.03	Car and van repairs and servicing	set26	
.04	Motor vehicle insurance and taxation	set26	
1041	Motor vehicle tax payments less refunds	set26	
.05	Petrol and diesel and other motor oils	set26	
.06	Other motoring costs	set26	
.062	Garage rent, motoring fines etc.	set26	
.11	Rail and tube fares	set26	
.12	Bus and coach fares	set26	
.13	Air and other travel and transport	set26	
14	Bicycles, boats - purchases and repairs	set26	
1143	Other non motorised vehicles, repairs &c	set26	
l 1p	Gross rent, mortgage int pay, water rates	set26	
124	TV, video, computers and audio equip	set26	
12411	TV, video, audio repairs, maintenence, parts	set26	
1242	Satellite dish - purchase + installation	set26	

## 1999-2000 derived variables that have been deleted for 2000-2001

Variable Description		Table Name	
t1249	Console computer games & computer softwa	set26	
t131	Cinema and theatre	set26	
t1310	Cash gifts, donations	set26	
t13104	Money sent abroad amount	set26	
t132	Sports admissions and subscriptions	set26	
t133	TV,video,satellite rental,TV licences	set26	
t1331	Television licences	set26	
t1335	Cable TV connection and subscription	set26	
t134	Miscellaneous entertainments	set26	
t135	Educational and training expenses	set26	
t1351	Education fees - school	set26	
t136	Hotel and holiday in UK	set26	
t137	Hotel and holiday abroad	set26	
t138	Other incidental holiday expenses	set26	
t139	Gambling payments	set26	
t1393	Lotteries (including National Lottery)	set26	
t13931	National lottery stakes & scratchcards	set26	
t13932	Other lotteries & scratchcards	set26	
t14	Repairs, maintenance and decorations	set26	
t141	Central heating repairs	set26	
t142	Child income - amount spent	set26	
t1421	House maintenance etc	set26	
t142t	CHILD INCOME - AMOUNT SPENT	set26	
t142t	Other payments recorded	set26	
t151	Life assurance, contrib to pension funds	set26	
t1511	Life assurance premiums	set26	
t1511	Contribs to pension/superntn ded frm pay	set26	
t1542	Income tax paid direct - amount	set26	
t1543	Self-employed: tax paid last 12 months	set26	
t1544	Tax dctd from pensions, invests, annuities	set26	
t1545	Tax paid on bonuses	set26	
t1547	Income tax received - amount	set26	
t155	National Insurance contributions	set26	
t1552	NI contribs paid by non-employees	set26	
t1552	Purchase or alteration dwellings, mortg	set26	
t1562	Capital repayment of mortgage	set26	
t15621	Capt repayment - building society mortg	set26	
t15622	Capt repayment - bunding society mortg	set26	
t15623	Capital repayment of Bank mortgage	set26	
t15624	Capital repayment of Bank mortgage  Capital repay Insurance Company mortg	set26	
	Central Heating installation		
t1563	House extensions	set26	
t1564	Savings and investments	set26	
t157	$\mathcal{C}$	set26 set26	
t159	Windfall receipts from gambling etc		
t21	Gas	set26	
t22	Electricity Other fuels	set26	
t23	Other fuels "Pareffin Color goe condles firewood etc."	set26	
t233	"Paraffin, Calor gas, candles, firewood etc	set26	
t311	"Meat pies, sausages and other meats	set26	
t3112	Offal and other uncooked meats	set26	
t312	"Fish, shellfish and fish products	set26	
t317	Milk products including cream	set26	
t32	"Pasta, rice, flour and other cereals	set26	
t320	"Potatoes, potato products (ex crisps)	set26	

Date amended: 23 August 2001 Page 3

## 1999-2000 derived variables that have been deleted in 2000-2001

/ariable	Description	Table Name	
321	Other vegetables	set26	
322	"Fruit, nuts	set26	
33	"Biscuits, cakes etc	set26	
330	"Fruit juices, squashes, bottled water	set26	
331	Biscuits including Kitkats.	set26	
334	Other convenience foods	set26	
3341	Dishes ready prepared	set26	
33413	Vegetarian foods	set26	
3342	Baby and diet foods	set26	
336	Restaurant meals	set26	
37	Take-away meals eaten at home	set26	
38	Other take-away food and snack food	set26	
381	Hot and cold food	set26	
382	Confectionary	set26	
383	Ice cream	set26	
384	Soft drinks	set26	
39	State school meals and meals at work	set26	
340	Other foods	set26	
3402	Other foods undefined	set26	
11	Hhold expenditure on beer, lager & cider	set26	
111	Beer and Lager	set26	
112	Cider	set26	
11off	Beer, cider (off-licence)	set26	
1on	Beer, cider (on licensed)	set26	
12	Hhold expenditure on wines & fortified w	set26	
121	Fortified wines	set26	
122	Non-fortified and still wines	set26	
23	Champagne and sparkling wines	set26	
2off	"Wines, fortified wines (off licence)	set26	
2on	"Wines, fortified wines (on licensed)	set26	
3	Hhold expenditure on spirits & liquers	set26	
4	Total expend on other alcoholic drinks.	set26	
41	Alcoholic soft drinks	set26	
42	Alcohol not otherwise specified	set26	
4off	Other alcoholic drinks bght at off liens	set26	
4on	Other alcoholic drnks bt on licned prems	set26	
2	Tobacco and other tobacco products	set26	
10	"Haberdashery, textiles and clothes hire	set26	
11	Footwear	set26	
59	"Ties, belts, hats gloves, etc	set26	
10	Pets and pet food	set26	
2	Floor coverings	set26	
3	Soft furnishings and bedding	set26	
74	"Gas and electric appliances, repairs	set26	
49	Repairs to gas & electricity appliances	set26	
5	"Kitchen/garden equip,hhold hardware	set26	
6	Kitchen and electrical consumables	set26	
62	Kitchen disposables inc. matches	set26	
8	Detergents and other cleaning materials	set26	
3	Telephone	set26	
31	Telephone account	set26	
34	Domestic help and child care	set26	
841	Domestic help inc service charge in rent	set26	
35	"Repairs to footwear, watches etc	set26	
36	"Laundry, cleaning and dyeing	set26	

## Family Expenditure Survey

# 2000-2001 : VOLUME 5, PART 5

## 1999-2000 derived variables that have been deleted for 2000-2001

Variable	Description	Table Name
:87	Subscriptions	set26
t88	Professional fees	set26
t881	Moving house	set26
t882	Legal fees excluding moving house	set26
t89	Other services	set26
t892	Bank and Post Office counter charges	set26
t893	"Second dwelling rent, council tax, water	set26
t899	Mortgage protection premiums	set26
t91	"Leather, travel goods, jewellery, watche	set26
t92	Baby toiletries and equipment	set26
t922	Baby Equipment	set26
t93	"Medicines, prescriptions and spectacles	set26
t94	"Medical, dental, optical, nursing fees	set26
t95	Toiletries and soap	set26
t96	Hair products and cosmetics	set26

Date amended: 23 August 2001 Page 5

		Table	
Variable	Description of Variable	Name	<b>Description of Change</b>

#### **Different bandings**

None found.

#### Variables that have been amended

d110210	Diary expenditure on school travel	Was in set900, is now in set14
d140402	Credit card interest payment	Was in set 24, is now in set900

#### Variables where the derivation condition/formula have changed.

0.50			
a053	Households with married women	set1	part of condition 1 changed
a053	Households with married women	set1	part of condition 9 changed
a054	Number of workers in household	set2	condition 1 changed
a056	Number of persons economically active	set2	condition 1 changed
a057	Men 65+ women 60+ economically active	set2	condition 1 changed
a058	Others - not economically active	set2	condition 1 changed
a117	No of new cars, vans bought last 12 months	set1	minor changes to condition 1
a117	No of new cars, vans bought last 12 months	set1	condition 2 changed
a117	No of new cars, vans bought last 12 months	set1	condition 3 changed
a117	No of new cars, vans bought last 12 months	set1	condition 4 changed
a118	No of second hand cars, vans bought lst 12 mths	set1	minor changes to condition 1
a118	No of second hand cars, vans bought lst 12 mths	set1	condition 2 changed
a118	No of second hand cars, vans bought lst 12 mths	set1	condition 3 changed
a118	No of second hand cars, vans bought lst 12 mths	set1	condition 4 changed
a1641	Satellite receiver in household	set1	new condition & formula
a170	Compact disc player in household	set1	new variable in formula
a206	Economic position	set41	a207 not used in condition 2
a206	Economic position	set41	a207 not used in condition 3
a206	Economic position	set41	variables changed in condition 4
a206	Economic position	set41	condition 5 changed
a206	Economic position	set41	condition 6 changed
a206	Economic position	set41	new variable in condition 7
a208	No of weeks away from work	set43	condition 1 & formula
a208	No of weeks away from work	set43	minor changes to condition 2
a208	No of weeks away from work	set43	condition 3 & formula changed
a266	Oap concessionary bus pass-type owned	set47	condition 2 formula changed
abi3	Private Medical Insurance.	setabi	condition 1 changed
abi4	Permanent Health Insurance (PHI).	setabi	condition 1 changed
			C
b150	Mortgage intrst\\prncpl - interest paid	set12	condition 1 & formula changed
b150	Mortgage intrst\\prncpl - interest paid	set12	conditions 2 & 3 not used
b159	Service charge included in rent - amount	set15	formula 2 amended
b166	Telephone, household share of acct (ex mobiles)	set15	formula 2 amended
b166	Telephone, household share of acct (ex mobiles)	set15	formula 3 amended
b1661	Mobile telephone account	set15	minor changes to condition 1
b1661	Mobile telephone account	set15	formula 2 amended
b1661	Mobile telephone account	set15	formula 3 amended
b1802	Bank & B/soc chgs - net amnt last 3 mths	set15	condition 1 changed
b1802	Bank & B/soc chgs - net amnt last 3 mths	set15	new condition 2.
b182	Credit purchase-net cost s-hand vehicles	set14	condition 1 changed
b182	Credit purchase-net cost s-hand vehicles	set14	formula 2 amended
	-		

Variable	Description of Variable	Table Name	Description of Change
b182	Credit purchase-net cost s-hand vehicles	set14	condition 3 & formula changed
b183	Credit purchase-net cost new vehicles	set14	condition 1 changed
b183	Credit purchase-net cost new vehicles	set14	formula 2 amended
b183	Credit purchase-net cost new vehicles	set14	condition & formula 3 adjusted
b191	Rent for TV\\Satellite\\VCR wkly amt	set20	condition 1changed
b192	Satellite subscription wkly amt	set20	condition 1 changed
b193	Cable subscription wkly amt	set20	condition 1 changed
b196	Life insurance before apr 84-amnt prmum	set24	condition 1 changed
b197	Life insurance after apr 84-amnt prmum	set24	condition 1 changed
b198	Endow policies after apr 84-amnt prmum	set12	condition 1 changed
b199	Endow policies before apr 84-amnt prmum	set12	condition 1 changed
b200	Mortgage intrst\\prncpl - last payment	set12	formula 1 amended
b200	Mortgage intrst\\prncpl - last payment	set12	minor change to condition no5
b200	Mortgage intrst\\prncpl - last payment	set12	minor change to condition no6
b200	Mortgage intrst\\prncpl - last payment	set12	minor changes to condition & formula 7
b200	Mortgage intrst\\prncpl - last payment	set12	minor change to condition no8
b201	Mortgage endow poly after 1983 amt prem	set12	condition 1 changed
b202	Mortgage endow ploy befor 1984 amt prem	set12	condition 1 changed
b203	Mixed mortgage - last payment	set12	formula 1 amended
b203	Mixed mortgage - last payment	set12	formula no2 amended
b203	Mixed mortgage - last payment	set12	minor change to condition no5
b203	Mixed mortgage - last payment	set12	minor amendment to formula no6
b203	Mixed mortgage - last payment	set12	minor amendment to formula no7
b206	Other insurance - total amount premium	set24	condition 1 changed
b208 b213	Mrtge protection pre apr 84-amnt prmum Mrtge protection after apr 84-amnt prmum	set12 set12	condition 1 changed
b213	Medical insurance - total amount premium	set24	condition 1 changed condition 1 changed
b260t	School meals - total amount paid last week	set24 set18t	condition 2 changed
b303	Pay-amount last time paid	set43	formula no1 - variable deleted
b312	Bonus earnings - gross amount	set43	formula no2 - tax rate changed
b330	Bonus earnings - net amount	set43	formula no2 - tax rate changed
b334	Money sent abroad - amount	set54	formula no1 - variable deleted
b342	Statutory maternity pay-amount received	set46	condition and formula no 3 changed
b342	Statutory maternity pay-amount received	set46	condition & formula 4 no longer used
b348	Pension(public\\private)last net payment	set47	condition & formula no2 changed
b388	Statutory sick pay-imputed gross amount	set49	minor change to condition no1
b388	Statutory sick pay-imputed gross amount	set49	formula no 4 - rate changed
b404	Disable wrking allw/Disable Pers Tax Cr.	set49	formula no 1 - variable changed
d010101u	Rent - diary expenditure	setunanonh	dqualif range in condition 1 changed
d010102	Mortgage instalment payment	set900	dqualif range in condition 1 changed
d010103	Mortgage endowment policy	set900	dqualif range in condition 1 changed
d010104	Mortgage protection policy	set900	dqualif range in condition 1 changed
d010105u	Council tax and rates - diary expenditure	setunanonh	dqualif range in condition 1 changed
d010106u	Water rates - diary expenditure	setunanonh	dqualif range in condition 1 changed
d010107	Service charges	set900	dqualif range in condition 1 changed
d010108	Structural insurance	set900	dqualif range in condition 1 changed
d010109	Contents insurance	set900	dqualif range in condition 1 changed
d010110 d010301	Ground rent Central heating installation(contractor)	set900 set900	dqualif range in condition 1 changed
d010301 d010302	Central heating installation(contractor) Central heating maintenance (contractor)	set900 set900	dqualif range in condition 1 changed dqualif range in condition 1 changed
d010302 d010303	•	set900 set900	
d010303 d010304	Capital improvements (contractor) Repairs,decor,replacements (contractor)	set900 set900	dqualif range in condition 1 changed dqualif range in condition 1 changed
d020101	Gas account payment	set900	dqualif range in condition 1 changed
d020102	Gas board budgeting payment	set900	dqualif range in condition 1 changed

Variable	Description of Variable	Table Name	Description of Change
d020201	Electricity account payment	set900	dqualif range in condition 1 changed
d020201	Electricity board budgeting payment	set900	dqualif range in condition 1 changed
d020302	Central heating oil	set900	dqualif range in condition 1 changed
d020304	Bottled gas	set900	dqualif range in condition 1 changed
d031302	State school meals	set900	dqualif range in condition 1 changed
d070101	Furniture inc beds and mattresses	set900	dqualif range in condition 1 changed
d070102	Soft floor coverings	set900	dqualif range in condition 1 changed
d080203	Telephone account	set900	dqualif range in condition 1 changed
d080206	Mobile phone account payments	set900	dqualif range in condition 1 changed
d080402	Bank service charges	set900	dqualif range in condition 1 changed
d080404	Conveyancing, est agents, surveyors fees	set900	dqualif range in condition 1 changed
d100101	New car or van outright purchase	set900	dqualif range in condition 1 changed
d100102	Second hand car outright purchase	set900	dqualif range in condition 1 changed
d100103	Car leasing payments	set900	dqualif range in condition 1 changed
d100104	New, second hand motor cycle purchase	set900	dqualif range in condition 1 changed
d100201	Car or van repairs and servicing	set900	dqualif range in condition 1 changed
d100204	Motor cycle repairs and servicing.	set900	dqualif range in condition 1 changed
d100401	Vehicle insurance	set900	dqualif range in condition 1 changed
d100402	Vehicle tax	set900	dqualif range in condition 1 changed
d110201	Railway and tube season tickets	set900	dqualif range in condition 1 changed
d110203	Bus and coach season tickets	set900	dqualif range in condition 1 changed
d110205	Combined fare season tickets	set900	dqualif range in condition 1 changed
d110210	Diary expenditure on school travel	set14	dqualif range in condition 1 changed
d110211	Water travel season ticket	set900	dqualif range in condition 1 changed
d130201	TV licences	set900	dqualif range in condition 1 changed
d130203	TV, video recorder, satellite TV rental	set900	dqualif range in condition 1 changed
d130301	Fees, maintenance: educational courses	set900	dqualif range in condition 1 changed
d130302	Fees, maintenance: non-household member	set900	dqualif range in condition 1 changed
d130303	Fees for leisure classes	set900	dqualif range in condition 1 changed
d130401 d130402	Holidays in UK: accomm (exc timeshares) Holidays abroad: accomm (exc timeshares)	set900 set900	dqualif range in condition 1 changed dqualif range in condition 1 changed
d130402	Timeshares, holiday homes overseas	set900	dqualif range in condition 1 changed
11.401.00	•	,000	
d140102	Private personal pension	set900	dqualif range in condition 1 changed
d140104	Money set aside for payment of bills	set900	dqualif range in condition 1 changed
d140201 d140202	Life,death insurance,non-house endowment Private medical insurance	set900	dqualif range in condition 1 changed dqualif range in condition 1 changed
d140202	Accident, sick, redundancy, other insurance	set900 set900	dqualif range in condition 1 changed
d140203	Money sent abroad	set900	dqualif range in condition 1 changed
d140304	Maintenance or separation allowance	set900	dqualif range in condition 1 changed
d140306	Money given to other household spenders	set900	dqualif range in condition 1 changed
d140308t	Pocket money to child no diary	set24t	dqualif range in condition 1 changed
d140309t	Money given to child for specific purpos	set24t	dqualif range in condition 1 changed
d140310t	Cash gifts to children	set24t	dqualif range in condition 1 changed
d140401	Credit card account payment	set900	dqualif range in condition 1 changed
d140402	Credit card interest payment	set900	dqualif range in condition 1 changed
d140403	Credit card ann standing charge payment	set900	dqualif range in condition 1 changed
d140404	Loan instalment payment	set900	dqualif range in condition 1 changed
d140405	Hire purchase instalment payment	set900	dqualif range in condition 1 changed
d140406	Club instalment payment	set900	dqualif range in condition 1 changed
d140501	Income tax payment	set900	dqualif range in condition 1 changed
u1 <del>1</del> 0501			
d140502	National insurance contribution	set900	dqualif range in condition 1 changed

Variable	Description of Variable	Table Name	<b>Description of Change</b>
fs113	Air & other travel & transport	set26a	two variables changed in formula
g021	Number of persons not economically active	set2	variable range in condition changed
p003	Gross wage,salary last time paid (main)	set43	variables used in formula no2 changed
p005	Gr wage, salary 1st time pd(main-13 wks)	set43	variable deleted from condition no1
p007	Normal gross wage,salary	set43	two new variables in formula no1
p007	Normal gross wage,salary	set43	formula no2 amended
p008	Normal gross wage,salary(13 wk rl)	set43	variable deleted from condition no1
p028	Soc-sec bnfts concurrent with earnings	set51	variable deleted from condition no1
p048	Income from investments	set46	variables changed in formula no1
p050	Income from other sources	set46	condition 2 and formula changed
p066	Paye tax refund, 13 week rule	set44	variable deleted from condition no1
p067	Income tax,payments less rfnds(13 wk rl)	set44	formual no1 amended
p068	Tax dctd from pensions, invests, annuities	set46	formula changed
p079	Income tax payments less refunds	set44	variables used in formula 1 amended
p081	Int & tax on bank,nat savings,TESSA,ISAs	set46	variables used in formula 1 amended
p108	New cars - purchase price	set54	condition no1 changed
p100	Second-hand cars -purchase price	set54	condition no1 changed
p109	Motor cycles cars - purchase price	set54	condition no1 changed
p110 p126	Miscellaneous - personal expenditure	set54	variable deleted from condition no1
p126t	Miscellaneous - personal expenditure	set54t	variable deleted from condition no1
•		set54t	
p126t	Miscellaneous - personal expenditure		dqualif range changed in condition 3 variable deleted from condition no1
p134	Fares etc (RPI)-personal expenditure	set54	
p134c	Fares etc (RPI)-personal expenditure	set54c	variable deleted from condition no1
p136 p184	Leisure services (RPI)- personal expend. Charitable gifts	set54 set44	minor changes to condition no1 variables changed in formula no1
•			· ·
p281u	Net rent, rates, coun tax, water, oth hsing	setunanonh	variables used in formula amended
p281p	Net rent,rates, coun tax, water, oth hsing	set12	variables used in formula amended
p389	Normal weekly disposable hhld income	set3	variables used in formula amended
p399	Disposable income + income in kind	set3	variables used in formula amended
p492p	index households	set1	wealthy cutoffs changed in condition1
p492p	index households	set1	wealthy cutoffs changed in condition 2
p492p	index households	set1	wealthy cutoffs changed in condition 3
p492p	index households	set1	wealthy cutoffs changed in condition 4
p492u	index households	setunanonh	wealthy cutoffs changed in condition 1
p492u	index households	setunanonh	wealthy cutoffs changed in condition 2
p492u	index households	setunanonh	wealthy cutoffs changed in condition 3
p492u	index households	setunanonh	wealthy cutoffs changed in condition 4
p526	Fares and other travel costs	set14	variable deleted from formula no1
p526c	Fares and other travel costs	set14c	variable deleted from formula no 1
p526t	Fares and other travel costs	set14t	variable deleted from formula no 1
p528t	Leisure services	set20	variable added to formula no 1
p530	Permanent second dwelling	set12	variable deleted from formula no 1
p530 p531	Other payments (neg acceptable)	set24	formula 1 amended
p531 p532	"Purchase of dwlng, imps and mort pymnt"	set24	variables used amended in formula 1
p532 p546	ONS fares and other travel costs	set24	variable deleted from formula no1
	ONS fares and other travel costs ONS fares and other travel costs	set24c	variable deleted from formula no1
p546c			variable deleted from formula no1
p546t	ONS fares and other travel costs	set24t	
p548	ONS leisure services	set24	formula 1 amended

Variable	Description of Variable	Table Name	Description of Change
540	Marillan	124	and the showed in Community and
p549	Miscellaneous	set24	variables changed in formula no1
p549t	Miscellaneous	set24t	variables changed in formula no 1
p551p	Total expenditure plus imputed values	set12	variable deleted from formula no1
p551tp	Total expenditure plus imputed values	set12t	variable deleted from formula no 1
p551tu	Total expenditure plus imputed values	setunanonht	variable deleted from formula no 1
p551u	Total expenditure plus imputed values	setunanonh	variable deleted from formula no 1
p553	Miscellaneous (national accounts)	set24	variable deleted from formula no 1
p553t	Miscellaneous (national accounts)	set24t	variable deleted from formula no 1
p559	Miscellaneous services - NAT ACCS	set24	formula largely rewritten
p565	Other travel - NAT ACCS	set14	1 variable deleted & 1added to formula
p565c	Other travel - nat accs	set14c	1 variable deleted & 1added to formula
p565t	Other travel - nat accs	set14t	Formula completely rewritten
wkgross	Gross pay - amount	setwrk	formula no1 rewritten
wkgross	Gross pay - amount	setwrk	variable deleted from formula no3

# Family Expenditure Survey

New expenditure codes

Expenditure Code	Description

None found.

**Family Expenditure Survey** 

1999-2000 Expenditure codes deleted for 2000-2001

Expenditure Code	Description of Expenditure Code	
There were no Expenditure	codes deleted.	

**Family Expenditure Survey** 

1999-2000 Expenditure codes deleted for 2000-2001

Expenditure Code	Description of Expenditure Code	
There were no Expenditure	codes deleted.	

1999-2000 Expenditure codes amended in 2000-2001

Expenditure Code	Description of Expenditure Code	Description of Change
030202	Other milk and cream	1 item added
030401	Butter	1 item added
030402	Margarine	1 item deleted and 1 item added
030508	Tinned and bottled meat and meat products	Description changed
030702	Processed potatoes and products	1 item added
030704	Processed and frozen vegetables	1 item added
030705	Pulses dried and processed	1 item added
030801	Fresh fruit	2 items added
030803	Dried fruit and nuts	1 item added
030903	Sweets and chocolates	1 item added
031002	Coffee	1 item added
031003	Food drinks	1 item added
031004	Fruit juice, drinks and squash	2 items added
031005	Carbonated drinks	1 item added
031006	Mineral water	1 item added
031107	Other convenience foods	3 items added
040206	Spirits, liquers	1 item added
040207	Alcoholic soft drinks, alcopops	4 items added
050101	Cigarettes	1 item added
070104	Bedding	1 item added
130207	Internet subscription fees	6 items added

## **Family Expenditure Survey**

**New tables for 2000-2001** 

Table Name	Description
busrfnd	Business refunds
ccint	Credit card interest
eout	Adult meals eaten out
mininc	Minimal income
youth	Young persons expenditure on eating out

Date amended: 26 September 2001

#### **Family Expenditure Survey**

1999-2000 Tables deleted for 2000-2001

Table Name	Description	
Raw tables		
impute	Imputation table used by SSD Editors is no longer required	

This diary table was deleted as the data is no longer required by users.

trav This table was used for data on travel to/from School which is no longer required.

winnings This diary table was deleted as the data is no longer required by users.

Derived tables

intcred

set26 This table was used in the preparation of Family Spending only.

set53t This table was used for the calculation of variable b397t - children's income, amount spent.

#### Changes in definitions 1991 to 2000-2001

#### 1991

No significant changes.

#### 1992

Housing – Imputed rent for owner occupiers and households in rent-free accommodation has been discontinued. For owner occupiers this was the rent they would have had pay themselves to live in the property they own, and for households in rent-free accommodation it was the rent they would normally have had to pay. Up to 1990 these amounts were counted both as income and as a housing cost. Mortgage interest payments were counted as a housing cost for the first time in 1991.

#### 199

Council Tax - Council Tax was introduced to replace the Community Charge in Great Britain from April 1993.

#### 1994-1995

New expenditure items - The definition of expenditure was extended to include two items previously shown under "other payments recorded". These are:

gambling payments; mortgage protection premiums.

Expenditure classifications - A new classification system for expenditures was introduced in April 1994. The system is hierarchical and allows more detail to be preserved than the previous system. New categories of expenditure were introduced and are shown in detail in table 7.1. The 14 main groups of expenditure were retained, but there were some changes in the content of these groups.

Gambling Payments - data on gambling expenditure and winnings are collected in the expenditure diary. Previously these have been excluded from the definition of household expenditure used in the FES. The data have been shown as memoranda items under the heading "Other payments recorded" on both gross and net bases. The net basis corresponds approximately to the treatment of gambling in the National Accounts. The introduction of the National Lottery has stimulated a reconsideration of this treatment. From April 1994, (gross) gambling payments have been included as expenditure in "Leisure Services". Gambling winnings continued to be noted as a memorandum item under "Other items recorded". They are treated as windfall income. They do not form a part of normal household income, nor are they subtracted from gross gambling payments. This treatment is in line with the PRODCOM classification of the Statistical Office of the European Communities (SOEC) for expenditure in household budget surveys.

Page 1

#### 1995-1996

 $Geographical\ overage\ -$  The FES geographical coverage was extended to mainland Scotland north of the Caledonian Canal.

*Under 16s diaries* - Two week expenditure diaries for 7-15 year olds were introduced following three feasibility pilot studies which found that children of that age group were able to cope with the task of keeping a two week expenditure record. Children are asked to record everything they buy with their own money but to exclude items bought with other people's money. Purchases are coded according to the same coding categories as adult diaries except for meals and snacks away from home which are coded as school meals, hot meals and snacks, and cold meals and snacks. Children who keep a diary are given a £5 incentive payment. A refusal to keep an under 16's diary does not invalidate the household from inclusion in the survey.

Pocket money given to children is still recorded separately in adult diaries; and money paid by adults for school meals and school travel is recorded in the Household Questionnaire. However, no attempt is made to determine whether money given to children for the purpose of school meals or school travel is actually used for those purposes.

Databases and tables in Family Spending reports for 1994-95 to 1997-98 excluded these expenditures, however, and pocket money and other transfers to children were shown in the same way as in earlier years. From 1998-99 they include results from the under 16s diaries. Appendix G of the 1998-99 report shows what difference the inclusion made.

#### 1996-1997

Self-employment - The way in which information about income from self-employment is collected was substantially revised in 1996-97 following various tests and pilot studies. The quality of such data was increased but this may have lead to a discontinuity. Full details are shown the Income Questionnaire, available from the address in the introduction.

Cable/satellite television - Information on cable and satellite subscriptions is now collected from the household questionnaire rather than from the diary, leading to more respondents reporting this expenditure.

 $Mobile\ phones$  - Expenditure on mobile phones was previously collected through the diary. From 1996/97 this has been included in the questionnaire.

Job seekers allowance (JSA) - Introduced in October 1996 as a replacement for Unemployment Benefit and any Income Support associated with the payment of Unemployment Benefit. Receipt of JSA is collected with NI Unemployment Benefit and with Income Support. In both cases the number of weeks a respondent has been in receipt of these benefits is taken as the number of weeks receiving JSA in the last 12 months and before that period the number of weeks receiving Unemployment Benefit/Income Support.

Retrospective recall - The period over which information is requested has been extended from 3 to 12 months for vehicle purchase and sale. Information on the purchase of car and motorcycle spare parts is no longer collected by retrospective recall. Instead expenditure on these items is collected through the diary.

State benefits - The lists of benefits specifically asked about was reviewed in 1996/97. See the Income Questionnaire for more information.

Sample stratifiers - New stratifiers were introduced in 1996/97 based on standard regions, socio-economic group and car ownership.

Government Office Regions - Regional analyses are now presented using the Government Office Regions (GORs) formed in 1994. Previously all regional analyses used Standard Statistical Regions (SSRs). For more information see Appendix F.

Page 3

#### 1997-1998

Bank/Building society service charges - Collection of information on service charges levied by banks has been extended to include building societies.

Payments from unemployment/redundancy insurances - Information is now collected on payments received from private unemployment and redundancy insurance policies. This information is then incorporated into the calculation of income from other sources.

Retired households - The definition of retired households has been amended to exclude households where the head of the household is economically active.

Rent-free tenure - The definition of rent-free tenure has been amended to include those households for which someone outside the household, except an employer or an organisation, is paying a rent or mortgage on behalf of the household.

National Lottery - From February 1997, expenditure on National lottery tickets was collected as three separate items: tickets for the Wednesday draw only, tickets for the Saturday draw only and tickets for both draws.

#### 1998-1999

Children's income – Three new expenditure codes were introduced: pocket money to children; money given to children for specific purposes and cash gifts to children. These replaced a single code covering all three categories.

Main job and last paid job - Harmonised questions were adopted.

#### 1999-2000

Working Family Tax Credit – the questionnaire was altered to deal with the replacement of Family Credit by Working Family Tax Credit in October 1999.

#### 2000-01

Household definition – the definition was changed to the harmonised definition which has been in use in the Census and nearly all other government household surveys since 1981. The effect is to group together into a single household some people who would have been allocated to separate households on the previous definition. The effect is fairly small but not negligible.

Up to 1999-2000 the FES definition was based on the pre-1981 Census definition and required members to share eating and budgeting arrangements as well as shared living accommodation. The definition of a household was:

One person or a group of people who have the accommodation as their only or main residence and (for a group)

share the living accommodation, that is a living or sitting room

and share meals together (or have common housekeeping).

The harmonised definition is less restrictive:

One person or a group of people who have the accommodation as their only or main residence and (for a group)

share the living accommodation, that is a living or sitting room

or share meals together or have common housekeeping.

The report on the 2000-01 survey, *Family Spending 2000-01*, contains a discussion of the effect of the change in Chapter 9. There is no clear-cut answer but the effect may be to increase average household size by a maximum of 1.9 per cent.

#### Question reductions

A thorough review of the questionnaire showed that a number of questions were no longer needed by government users. These were cut from the 2000-01 survey to reduce the burden on respondents. The reduction was fairly small but it did make the interview flow better. All the questions needed for a complete record of expenditure and income were retained.

#### Redesigned diary

The diary was redesigned to be easier for respondents to keep and to look cleaner. The main change of substance was to delete the column for recording whether each item was purchased by credit, charge or shop card.

Page 5

Ending of MIRAS

Tax relief on interest on loans for house purchase was abolished from April 2000. Questions related to MIRAS were therefore dropped. They included some that were needed to estimate the amount if the respondent did not know it. A number were retained for other purposes, however, such as the amount of the loan still outstanding which is still asked for households paying a reduced rate of interest because one of them works for the lender.

Page 6

Date produced : 14 September 2001

#### **QUESTIONNAIRE CHANGES FROM APRIL 2000**

Based on a paper by Social Survey Division

This paper lists the changes in the content of the FES questionnaires from April 2000. It covers new questions, deletions, amendments to wording, coding changes within specific questions on the Household and Income questionnaires and major routing changes.

The first column gives the question reference number used in the 2000-2001 questionnaire and the second column gives the question reference number used in the 1999-2000 questionnaire. If this number is pre-fixed by an 'H' then the question can be found in the Household Questionnaire and if it is pre-fixed by an 'I' it can be found in the Income Questionnaire.

The third column gives the 2000-2001 question text and explains the nature of the change; in the case of completely new questions a routing statement is included, e.g. APPLIES IF BUYING WITH MORTGAGE, followed by the question text. Changes to coding or wording are indicated by "CODING CHANGE" or "REWORDED" followed by the new codes or wording.

Questions are covered by topic area, e.g. mortgages.

The full text and routing for all questions and the diary coding frame are available in separate documentation, see Volume 6 - Variable Relationships.

#### Changes to the questionnaire in 2000-2001

The main changes to the FES questionnaire for 2000-2001 are summarised below.

Date Produced: 20 September 2001 Page 1

#### **MAIN QUESTIONNAIRE CHANGES**

#### **Household Questionnaire**

- 1. Changes to coding frame for Standard Statistical Regional stratifier.
- 2. New questions for household definitions.
- 3. New questions for highest income household member.
- 4. New questions for consumer durables.
- 5. Replacement questions regarding credit card information

#### **Income questionnaire**

- 1. New questions for employee pay details
- 2. New questions for employee pay details (Minimum income)

Page 2 Date Produced: 20 September 2001

2000-2001 1999-2000 question question

number number 2000-2001 Questionnaire text or description of change

#### **Household Composition**

# **Household Questionnaire**

#### **Administrative variables**

#### TABLE: HHOLD1 Coding frame amended

H_10.92	N/A	FESREG (*)	Standard Statistical Regional stratifier
		Variable attached	at SSR level
		Northern Metropol	itan1
		North Non-Metrop	olitan2
		Yorks & Humbersi	de Metropolitan3
			de Non-Metropolitan4
			politan5
			letropolitan6
		East Midlands	7
		West Midlands Me	etropolitan8
		West Midlands No	n-Metropolitan9
		East Anglia	
		Greater London No	orth East11
		Greater London No	orth West12
		Greater London So	outh East13
		Greater London So	outh West14
		South East Metrop	olitan15
		South East Other	16
		South West	
		Wales I	18
		Wales II	
		Scotland I	20
		Scotland II	21
		Scotland III Metro	politan22
		Scotland III Non-M	Metropolitan23
		Scotland IV	24
		Northern Ireland	25

2000-2001	1999-2000	
question	question	
number	number	2000-2001 Questionnaire text or description of change

#### **NEW QUESTIONS FOR H/H DEFINITIONS**

H_20.37	NA	QTHCOMP. <b>OldHhDef</b>	
		$APPLIES\ IF\ NumAdult = >1$	
		Do all household members share the living accommodation AND shar meals together, or have common housekeeping arrangements? THIS IS THE OLD FES DEFINITION	re
		Yes1	
		No2	
H_20.38	NA	QTHComp.NumHhDef  APPLIES IF OldHhDef = No  How many groups are there within the household who share living accommodation AND share meals together/common housekeeping?  THAT IS, THE OLD FES HOUSEHOLD DEFINITION WITHIN THIS HARMONISED DEFINITION (A GROUP CAN CONSIST OF ONE PERSON)  210	

# NEW QUESTIONS FOR HIGHEST INCOME HOUSEHOLD MEMBER

H_20.51	NA	QHOH.HiHNum  APPLIES IF HHldr = 3  You have told me that [Person] jointly own or rent the accommodation.  Which of you/who, has the the highest income (from earnings, benefits, pensions and any other sources?  Person number of HiH  114.
H_20.52	NA	IF TWO OR MORE HAVE SAME INCOME, ENTER 11  QHOH.JntEldA
		APPLIES IF HiHNum = 11 Enter person number of the eldest joint householder from those with the same highest income. ASK OR RECORD 114
H_20.53	NA	QHOH.JntEldB  APPLIES IF (HiHNum = DK) OR (HiHNum = RF)  Enter person number of the eldest joint householder  ASK OR RECORD  114

2000-2001 1999-2000 question question

number number 2000-2001 Questionnaire text or description of change

#### **Accommodation**

#### **TABLE: ADDRESS**

 $H_50.40$  to  $H_50.55$  inclusive have been deleted.

#### **Services in Rent**

#### **TABLE: HBSERVAM**

H 70.100 has been deleted

#### **TABLE: RENTS**

H\_70.115 to H\_70.150 inclusive have been deleted.

H\_70.160 to H\_70.165 inclusive have been deleted.

#### **Mortgages**

#### **TABLE: MORT**

H\_110.10 to H\_110.35 inclusive have been deleted.

H\_110.45 has been deleted.

H\_110.55 has been deleted.

H\_110.90 to H\_110.100 inclusive have been deleted.

 $H\_110.150$  to  $H\_110.175$  inclusive have been deleted.

#### **TABLE: MORTPROT**

H\_120.15 has been deleted.

#### **TABLE: MORT**

H\_130.10 has been deleted

#### **Private Pension Contributions**

#### TABLE: HHOLD1

 $H_160.35$  to  $H_160.40$  inclusive have been deleted.

#### **Questionnaire Changes**

2000-2001	1999-2000	
question	question	
number	number	2000-2001 Questionnaire text or description of change

#### Life Insurance

#### **TABLE: LIFE**

H\_170.10 has been deleted.

#### **Consumer Durables**

#### TABLE: HHOLD2

H\_190.30 has been deleted.

H\_190.60 has been deleted.

#### NEW QUESTIONS FOR CONSUMER DURABLES

H\_190.80 NA QDURABLE.Access (Access0..Access4)

APPLIES IF INTERNET = 1

How does your household access the internet at home?

CODE ALL THAT APPLY

Home Computer1Digital Television2Mobile Phone3Games console4Other5

Page 6

Date Produced: 20 September 2001

2000-2001 question number	1999-2000 question number	2000-2001 Questionnaire text or description of change
H_190.85	NA	QDURABLE. <b>WWWeb</b> (*)  APPLIES IF INTERNET = I
		May I just check, are you able to access the World Wide Web via your home internet connection?
		Yes
		Telephone Payments

#### TABLE: PHONE

H\_200.80 to H\_200.95 inclusive have been deleted.

# **Electricity Payments**

#### **TABLE: HHOLD2**

H\_210.75 to H\_ 210.90 inclusive have been deleted.

#### **Gas Payments**

#### **TABLE: HHOLD2**

H\_220.80 to H\_220.95 inclusive have been deleted.

# TV, video, satellite rental and subscription

TABLE: TV		Reworded: question now reads:	
H_240.20	H_240.20	QBTV.Num. <b>TVType</b> (*)	
		What does your payment cover?	
		Hire of a TV, video, satellite subscription	1
		Subscription to satellite channels only	. 2
		Cable TV including telephone	. 3
		Cable TV not including telephone	4

#### 2000-2001 : VOLUME 5, PART 13

**Questionnaire Changes** 

**Family Expenditure Survey** 

2000-2001 question

number

1999-2000 question number

2000-2001 Questionnaire text or description of change

#### **TV Licences**

#### TABLE: HHOLD2

Reworded: question now reads:

H\_250.5 H\_250.5

QBTV.TVLice (\*)

APPLIES TO HOH OR WIFE

How much was your last TV licence payment?

APPLIES IF QDURABLE.TVCOL = 1,2 OR QDURABLE.TVBW = 1,2 IF RESPONDENT HAS NO TV AND HAS NOT ENTERED ANY PAYMENTS ENTER 0. UP TO 1st APRIL 2000; ANNUAL COLOUR TV: £101.50, B/W: £33.50 RESPONDENTS MAY PAY MORE OR LESS IN

CERTAIN CIRCUMSTANCES.

0..997.00

#### **Owned and leased vehicles**

#### **TABLE: VOWNED**

H\_260.30 has been deleted.

#### **Vehicles bought**

#### **TABLE: VEHICLE**

 $H_270.30$  and  $H_270.35$  have been deleted.

#### Vehicles sold

#### **TABLE: VEHICLE**

H\_280.30 has been deleted.

# Company/ continuous use vehicles

#### **TABLE: VEHICLE**

H 290 .30 has been deleted.

2000-2001 1999-2000
question question
number 2000-2001 Questionnaire text or description of change

TABLE: VUSED Reworded: question now reads:

 $APPLIES\ IF\ QTVEH4.Num.Band{=}DK$ 

Please give the registration letter Y

#### **Vehicle servicing**

#### **TABLE: SERV**

H\_310.50 has now been deleted.

#### **Credit cards**

**TABLE: HHOLD2** 

#### REPLACEMENT QUESTIONS FOR CC

TABLE: CC (April 2000 – September 2000)

Section records details of up to 5 credit cards. Applies if QCREDIT.AnnChg = 1.

2000-2001	1999-2000	
question	question	
number	number	2000-2001 Questionnaire text or description of change

# **Credit cards**

TABLE: HHO	LD2	
H_ 340.5	NA	QCREDIT.CreditCd (*)  APPLIES TO ALL SPENDERS  Do (any of) you AT PRESENT have a credit card, charge card, shop card or store card account on which interest can be charged or on which an annual standing charge is made?  Yes
H_340.10	NA	QCREDIT.Annchg (*)  APPLIES IF QCREDIT.CreditCd = 1  Have (any of) you paid an annual standing charge for any of these cards in the 12 months since (date)?  Yes
TABLE: CC		
H_340.15	NA	QCREDIT.Num.Person (*) Card (number)/Who paid the annual standing charge for this card? 1 14
H_ 340.20	NA	QCREDIT.Num.CardNum (*) Card number 15
H_340.25	NA	QCREDIT.Num.ChgAmt (*) Card (number) /How much have you paid in total in the last 12 months since (date) 1997
H_340.30	NA	QCREDIT.Num. <b>DVChg</b> (*) (DV FOR QCREDIT.Num.ChgAmt) 0 997.00
H_340.35	NA	QCREDIT.Num.CardBrnd (*)  APPLIES IF QCREDITC.ChgAmt = DON'T KNOW  Card (number)/What is the name of the card?  (Text field)
H_340.40	NA	QCREDIT.Num.AnyMore (CC_AnyMore)  Have (any of) you paid an annual standing charge for any more of these cards in the 12 months since (date)?  Yes

2000-2001 1999-2000 question question

number 2000-2001 Questionnaire text or description of change number

#### TABLE: CC (October 2000 – March 2001)

#### **Credit cards**

TABLE: HHO	OLD2	
H_ 340.5	NA	QCREDIT.CreditCd (*)  APPLIES TO ALL SPENDERS  Do (any of) you AT PRESENT have a credit card, charge card, shop card or store card account on which interest can be charged or on which an annual standing charge is made?  Yes
H_340.10	NA	QCREDIT.Annchg (*)  APPLIES IF CreditCd = 1  Have (any of) you paid an annual standing charge for any of these cards in the 12 months since (date)?  Yes
H_340.16	NA	QCREDIT.Paycard  Have (any of) you paid a credit/charge/shop/store card account during the last month?  Yes
H_340.21	NA	QCREDIT.Interest  APPLIES IF Paycard = 1  Was any interest shown on the account (s)?  Yes
TABLE: CC		
H_ 340.25	NA	QCREDIT.Num.ChgAmt (*)  APPLIES IF AnnChg = 1  Card (number) /How much have you paid in total in the last 12 months since (date)?

1..997

# **2000-2001 : VOLUME 5, PART 13**

•	•	$\alpha$
Question	naire	Changes

Family Expenditure Survey

2000-2001 question number	1999-2000 question number	2000-2001 Questionnaire text or description of change
H_340.30	NA	QCREDIT.Num. <b>DVChg</b> (*) (DV FOR QCREDIT.Num.ChgAmt) 0. 997.00
H_ 340.35	NA	QCREDIT.Num.CardBrnd (*)  APPLIES IF QCREDITC.ChgAmt = DON'T KNOW  Card (number)/What is the name of the card?  (Text field)
H_ 340.40	NA	QCREDIT.Num.AnyMore (CC_AnyMore)  APPLIES IF CHGAMT = RESPONSE  Have (any of) you paid an annual standing charge for any more of these cards in the 12 months since (date)?  Yes
TABLE: CO	CINT	
H_340.45	NA	Index variable: (CCInt_CardNum) Card number 15
H_340.50	NA	QCREDIT.Num. <b>Person</b> (*)  APPLIES IF INTEREST = 1  Card (number) /Who paid the annual standing charge for this card?  114
H_340.55	NA	QCREDIT.IntAmt  How much interest was shown?  (Not total account payment)  0.01997.00
H_340.60	NA	QCREDIT. <b>DVINT</b> (*) (DV FOR INTAMT) 0997.00
H_340.65	NA	QCREDIT.Anymore  APPLIES IF INTAMT = RESPONSE  Have you paid any more of these card accounts on which interest was shown in the last month?  Yes

2000-2001	1999-2000	
question	question	
number	number	2000-2001 Questionnaire text or description of change

#### **Formal loans**

TABLE: LO	ANDEA	Reworded: question now reads:	
H_ 350.10	H_350.10	QLOANSA.Num.Org (LOANDEA_Org) Loan (number) /Is the source of the loan a finance house, or credit union, second mortgage or a per	sonal loan from
		bank or building society	1
		DSS Social Fund or repayment of a student loan?	3

# Formal loans: items obtained

#### TABLE: ITEMDEA

H\_350.90 has been deleted.

H\_350.100 has been deleted.

#### 2000-2001 : VOLUME 5, PART 13

**Questionnaire Changes** 

**Family Expenditure Survey** 

2000-2001 question

number

1999-2000 question number

2000-2001 Questionnaire text or description of change

#### Hire purchase - instalments being paid

#### **TABLE: AGREE1**

H\_360.25 has been deleted

H\_360.35 has been deleted

H\_360.65 and H\_360.70 have been deleted.

#### Hire purchase - no instalments paid

#### TABLE: HP

 $H_370.30$  and  $H_370.35$  have been deleted.

#### Club credit

#### **TABLE: LASTPAY**

H\_380.15 has been deleted.

#### Club credit - items obtained

#### **TABLE: LASTMTH**

H\_380.55 has been deleted

#### **Central heating installation**

#### TABLE: CHI

 $H_{400.20}$  has been deleted.

#### House maintenance

#### TABLE: DEC

H\_430.70 to H\_430.80 inclusive have been deleted.

2000-2001 1999-2000 question question

number number 2000-2001 Questionnaire text or description of change

#### **Moving house fees**

#### TABLE: HEXP2

H\_440.55 has been deleted.

#### Package holidays

#### **TABLE: HOLPACK**

 $H_480.30$  and  $H_480.35$  have been deleted.

H\_480.90 to H\_480.110 inclusive have been deleted.

#### **Bank charges**

#### TABLE: BK

H\_520.6 to H\_520.10 inclusive have been deleted.

#### **School Travel**

#### **TABLE: HHOLD2**

H\_ 570.5 has been deleted.

TABLE: TRAV (incorporating questions H\_570.10 to H\_570.35 inclusive) has all been deleted

#### Leisure classes

#### **TABLE: EDF2**

H\_610.30 has been deleted.

#### **Separation/ maintenance allowance**

#### TABLE: SEP

H\_630.35 has been deleted.

#### 2000-2001 : VOLUME 5, PART 13

**Questionnaire Changes** 

**Family Expenditure Survey** 

2000-2001 question

number

1999-2000 question

number 2000-2001 Questionnaire text or description of change

#### **Income questionnaire**

#### **Employment Status (b)**

#### TABLE: ILO

I\_10.105 to I\_10.115 inclusive have been deleted.

#### **Employment Status (c)**

#### TABLE: ILO

I\_10.135 to I\_10.155 inclusive have been deleted.

I\_10.165 to I\_10.180 inclusive have been deleted.

### Job Description (a)

#### **TABLE: JOBMAIN**

 $I_20.25$  and  $I_20.30$  have been deleted.

#### **Job Description (c)**

#### TABLES: JOB2/JOB3

I\_30.5 and I\_30.6 have been deleted.

 $I_30.15$  to  $I_30.25$  inclusivehave been deleted.

I\_30.30 to I\_30.35 inclusive have been deleted.

2000-2001	1999-2000
question	question

number number 2000-2001 Questionnaire text or description of change

### Employee pay details (a)

TABLE: PAYMAIN		Reworded: question now reads:
I_50.80	I_50.80	QTEmpPay. <b>DedTot</b> (*)  APPLIES IF deductions for charities (QTEmpPay.DedCh = 1) <b>How much was deducted for charities?</b>
		0.01997.00

I\_50.85 to I\_50.105 inclusive have been deleted.

I\_50.270 and I\_50.275 have been deleted.

#### NEW QUESTIONS FOR EMPLOYEE PAY DETAILS

I_50.281	NA	QTEmpPay. <b>PayCrdt</b> (*)  APPLIES IF (Workstat = 1) AND PayAmt = Response  Did your take home pay include any of the following?  Working Families Tax Credit
I_50.282	NA	QTEmpPay.WFTCAmt (*)  APPLIES IF PayCrdt = 1  How much was included for Working Families Tax Credit?  0.01997.00
I_50.283	NA	QTEmpPay. <b>WFTCPer</b> (*)  APPLIES IF WFTCAmt = Response  What period did this cover?  SEE APPENDIX A: STANDARD PERIOD CODES
I_50.284	NA	QTEmpPay. <b>DVWFTC</b> APPLIES IF WFTCPer = Response <b>DV for amount of Working Families Tax Credit</b> 0.01997.00

# **2000-2001 : VOLUME 5, PART 13**

•	•	$\alpha$
Question	ınaıre	Changes

Family Expenditure Survey

2000-2001 question number	1999-2000 question number	2000-2001 Questionnaire text or description of change
I_50.285	NA	QTEmpPay. <b>DPTCAmt</b> (*)  APPLIES IF PayCrdt = 2 <b>How much was included for Disabled Person's Tax Credit?</b> 0.01997.00
I_50.286	NA	QTEmpPay. <b>DPTCPer</b> (*)  APPLIES IF DPTCAmt = Response  What period did this cover ?  SEE APPENDIX A: STANDARD PERIOD CODES
I_50.287	NA	QTEmpPay. <b>DVDPTC</b> APPLIES IF DPTCPer = Response <b>DV for amount of Disabled Person's Tax Credit</b> 0.01997.00
I_ 50.288	NA	QTEmpPay.CCTCrdt(*)  APPLIES IF PayCrdt = 1,2  Do either of these payments include a Child Care Tax Credit to help pay for child care expenses?  Yes

I\_50.310 has been deleted.

 $I_50.325$  and  $I_50.330$  have been deleted.

1999-2000
question

number number 2000-2001 Questionnaire text or description of change

# NEW QUESTIONS Employee pay details (Minimal Income questions)

#### **TABLE: MININC**

Section applies if PAYDAT in last 12 months and full income information cannot be obtained.

I_50.430	NA	QTEmpPay.Person (*) Person number – i.e who receives income ? 114
I_50.431	NA	QTEmpPay.incnum(*) Income number 14
I_50.435	NA	QTEmpPay.MinInc.Source(*) Income source (record description or code letters) ? {Source is text field, for use with Prompt Card Z}
I_50.440	NA	QTEmpPay.GrosAmt(*) What was the GROSS amount paid (before any deductions), the last time it was paid? SELF EMPLOYED: What was your income from this/job business for the most recent period for which you have figures? (Gross = after expenses deducted, Net = after tax and National Insurance deducted)  0.0199997.00, DK, refusal.
I_50.445	NA	QTEmpPay.GrosPer(*) What period did this cover ? SEE APPENDIX A, STANDARD PERIOD CODES
I_50.450	NA	QTEmpPay.NetAmt(*) What was the NET amount paid the last time it was paid? 0.0199997.00, DK, refusal
I_50.455	NA	QTEmpPay.NetPer What period did this cover? SEE APPENDIX A, STANDARD PERIOD CODES

#### 2000-2001 : VOLUME 5, PART 13

Ouesti	onnaire	Changes
Oucsu	Ullian C	Changes

**Family Expenditure Survey** 

2000-2001 question number	1999-2000 question number	2000-2001 Questionnaire text or description of change
I_50.460	NA	QTEmpPay. Anymore
		Do you have anymore incomes ?
		Yes1
		No
		Income from Self-employment (a)

#### **TABLE: SEJOB**

I\_60.85 has been deleted.

I\_60.245 and I\_60.250 have been deleted.

### Income as mail order agent or baby sitter

**TABLE: VARIOUSI** Questions 100.5 to 100.25 have been deleted.

(questions incorporated into odd job questions)

#### **Income from odd jobs** Reworded question now reads

QTOddJ: Section applies to ALL SPENDERS

In the 12 months since (date), have you received any money for odd jobs, including mail\_order agent work or babysitting work, or any occasional fees for work or professional advice that we have not yet covered?

#### **Redundancy payments**

#### **TABLE: VARIOUSI**

I\_120.20 and I\_120.25 have now been deleted

Page 20 Date Produced: 20 September 2001

	penditure Surv	ey Questionnaire Cha
2000-2001 question	1999-2000 question	
number	number	2000-2001 Questionnaire text or description of change
		Concessionary bus travel
TABLE: VA	ARIOUSI	Reworded: question now reads:
I_ 130.10	I_130.10	QTBusPas. <b>BTType</b> (*)  APPLIES IF QTBusPas.BusPass = 1
		Is it a
		Free bus pass or permit(1)
		half-fare bus pass or permit(2)
		flat-fare bus pass or permit(3)
		bus tokens or bus tickets(4)
		London: Bus or tube(5)
		London: Bus, tube or rail(6)
		or any other type of concessionary OAP bus pass? (7)
		Benefits (b)
TABLE: BE	ENR2	
		50.39 have been deleted.
I_150.29, I_	_150.34 and I_1	
I_150.29, I_	_150.34 and I_1	Regular allowances
I_150.29, I_  TABLE: VA		Regular allowances
TABLE: VA	ARIOUSI	Regular allowances we have been deleted.
TABLE: VA	ARIOUSI	
TABLE: VA	ARIOUSI I_230.30 inclusiv	ve have been deleted.
<b>TABLE: V</b> AI_230.20 to	ARIOUSI I_230.30 inclusiv	we have been deleted.  Money sent abroad

 $I\_\,280.20$  and  $\,I_{-}\,280.25$  have been deleted.

#### 2000-2001 : VOLUME 5, PART 13

**Questionnaire Changes** 

**Family Expenditure Survey** 

2000-2001	1999-2000
question	question

number number 2000-2001 Questionnaire text or description of change

#### **Interest on Investments - TESSAs**

#### **TABLE: VARIOUSI**

I\_290.10 has been deleted

 $I_290.35$  to  $I_290$  .45 inclusive have been deleted

#### Interest on Investments - Building Society/ Bank accounts

#### **TABLE: VARIOUSI**

I\_300.5 to I\_300.30 inclusive have been deleted.

#### **Interest on Investments - Other bank or society accounts**

#### **TABLE: VARIOUSI**

I\_340.5 to I\_ 340.20 inclusive have been deleted.

#### **Children's income**

#### **TABLE: CHILDINC**

I\_360.25 has been deleted

# Raw Variable Changes for 2000-2001

# Derived Variable Changes for 2000-2001

# **Expenditure Code Changes** for 2000-2001

# Other Changes for 2000-2001

Tue Oct 3 15:50:13 2000

filename: newyear2000/deramend2000

filename: Derived amended in 2000-2001.xls

This output is used to produce PART of Section 5, Part 6

Previous year derived variables that have been amended in the current year.

			condition		
variable	Variable description	table name	number	condition	formula
a054	Number of workers in household	set2	1	a206 in (1,2,3,7)	count(a206)
0.7.4	Number of persons economically				(200)
a056	active Men 65+ women 60+	set2	1	a206 between 1 and 5	count(a206)
0.57				206	v/ 200
a057	economically active	set2	1	a206 = 6	count(a206)
a058	Others - not economically active	set2	1	a206 = 7  or  a206 = 0	count(a206)
	No of new cars, vans bought last				
a117	12 mnths	set1	1	vbought_vehtype in (1,2) and nocarp = 1	count(vbought_vehnum)
				itemdea_pabl0001 = '100105' and (startdat - (date('15/' +	
	No of new cars, vans bought last		_	varchar(loandea_loandatm) + '/' + varchar(loandea_loandaty))) <= '12 months') and	
a117	12 mnths	set1	2	itemdea_loannum = loandea_loannum	avg(a117) + count(itemdea_pabl0001)
	No of new cars, vans bought last			agree1_pabh0011 = '100105' and startdat - (date('15/' + varchar(hpm) + '/' +	
a117	12 mnths	set1	3	varchar(hpy))) <= '12 months'	avg(a117) + count(agree1_pabh0011)
117	No of new cars, vans bought last		4	agree2_pabh0001 = '100105' and startdat - (date('15/' + varchar(hpdatm) + '/' +	( 117)
a117	12 mnths	set1	4	varchar(hpdaty))) <= '12 months'	avg(a117) + count(agree2_pabh0001)
-110	No of s-hand cars, vans bought lst 12 mnths	41	1	-h	
a118	12 mntns	set1	1	vbought_vehtype in (1,2) and nocarp = 2 itemdea_pabl0001 = '100106' and (startdat - (date('15/' +	count(vbought_vehnum)
	No of s-hand cars, vans bought lst			varchar(loandea_loandatm) + '/' + varchar(loandea_loandaty))) <= '12 months')	
a118	12 mnths	set1	2	and itemdea_loannum = loandea_loannum	avg(a118) + count(itemdea_pabl0001)
a110	No of s-hand cars, vans bought lst	SCLI		agree1_pabh0011 = '100106' and startdat - (date('15/' + varchar(hpm) + '/' +	avg(a110) + count(nemuea_paoiooo1)
a118	12 mnths	set1	3	varchar(hpy))) <= '12 months'	avg(a118) + count(agree1_pabh0011)
u110	No of s-hand cars, vans bought lst	5011	3	agree2_pabh0001 = '100106' and startdat - (date('15/' + varchar(hpdatm) + '/' +	a. S(arro)   count(agreer_paonoorr)
a118	12 mnths	set1	4	varchar(hpdaty))) <= '12 months'	avg(a118) + count(agree2_pabh0001)
		5001			a. O(a a) sam(aB. co=_bachooox)
a1641	Satellite receiver in household	set1	1	(satcab0 = 1  or  satcab1 = 1  or  satcab2 = 1  or  satcab3 = 1)	1
a170	Compact disc player in household	set1	1		cdplay
	Economic position	set41	2	a200 = 12 and $a220 >= 31$ and $a204 != 1$	2.
	Economic position	set41		a200 = 12 and $a220 = 31$ and $a204 = 1$	3
	Economic position	set41		a200 in (3,4)	4

variable	Variable description	table name	condition number	condition	formula
a206	Economic position	set41	5	a204 = 1	5
a200	Leonomic position	30141	3	((a004 = 1  and  a005 >= 65)  or  (a004 = 2  and  a005 >= 60))  and  a200  between  5	3
a206	Economic position	set41	6	and 7	6
a206	Economic position	set41	7	a200 > 0 and $a206 = 0$	7
a208	No of weeks away from work	set43	1		interval ('weeks',startdat-dtjbl)
	Oap concessionary bus pass-type				, ,
a266	owned	set47	2	buspass = 3	9
abi3	Private Medical Insurance.	setabi	1	minstype = 1	sum(dvmins)
	Permanent Health Insurance				
abi4	(PHI).	setabi	1	minstype = 2	sum(dvmins)
	Mortgage intrst\\prncpl - interest				
b150	paid	set12	1	morttype in (2,5)	dvmort * 0.65
	Service charge included in rent -				
b159	amount	set15	2	ownrent in $(3,4)$ and hbstmt = 1 and rent > 0	sum(dvhbsv)
	Bank & B/soc chgs - net amount				
b1802	last 3 mths	set15	1	buspurp = 1	sum (((100 - busperc) * dvsrv) / 100.0)
	Bank & B/soc chgs - net amount		_		
b1802	last 3 mths	set15	2		avg(b1802) + sum(dvsrv)
	Credit purchase-net cost s-hand			itemdea_pabl0001 = '100106' and (startdat - (date('15/' + varchar(loandea_loandatm) + '/' + varchar(loandea_loandaty))) <= '3 months') and itemdea_loannum =	
b182	vehicles	set14	1	+ varchar(toandea_toandaty))) <= 3 months ) and itemdea_toannum = loandea_toannum	sum ((itemdea_lcashp - itemdea_lpartex) / 13.0)
0102	Credit purchase-net cost s-hand	30114	1	agree1_pabh0011 = '100106' and startdat - (date('15/' + varchar(hpm) + '/' +	sum ((nemaca_reasnp = nemaca_rpartex) / 13.0)
b182	vehicles	set14	2	varchar(hpy))) <= '3 months'	avg(b182) + sum((cashhp)/13.0)
0102	Credit purchase-net cost s-hand	50111		agree2_pabh0001 = '100106' and startdat - (date('15/' + varchar(hpdatm) + '/' +	a.8(0102) + ban((cabinip)) 13.0)
b182	vehicles	set14	3	varchar(hpdaty))) <= '3 months'	avg(b182)+ sum((agree2_cashprce - partex) / 13.0)
	Credit purchase-net cost new		_	agree1_pabh0011 = '100105' and startdat - (date('15/' + varchar(hpm) + '/' +	Name of the second seco
b183	vehicles	set14	2		avg(b183) + sum((cashhp) / 13.0)
	Credit purchase-net cost new			agree2_pabh0001 = '100105' and startdat - (date('15/' + varchar(hpdatm) + '/' +	
b183	vehicles	set14	3	varchar(hpdaty))) <= '3 months'	avg(b183) + sum((agree2_cashprce - partex) / 13.0)
	Credit purchase-net cost new			itemdea_pabl0001 = '100105' and (startdat - (date('15/' + varchar(loandea_loandatm) + '/'	
b183	vehicles	set14	1	+ varchar(loandea_loandaty))) <= '3 months') and itemdea_loannum = loandea_loannum	sum((itemdea_lcashp - itemdea_lpartex) / 13.0)
1.101	Rent for TV\\Satellite\\VCR wkly	20	_		(1)
b191	amt	set20	1	tvtype = 1	sum(dvtv)
h102	Catallita subscription vuldy and	nat20	1	tutuma — 2	ours (duty)
b192 b193	Satellite subscription wkly amt  Cable subscription wkly amt	set20 set20	1	tvtype = 2 tvtype in (3,4)	sum(dvtv) sum(dvtv)
0193	Life insurance before apr 84-amnt	setzu	1	ivtype iii (3,4)	Suin(uviv)
b196	prmum	set24	1	linstype in $(1,3,4,5)$ and linsstyr = 1	sum(dvlins)

variable	Variable description	table name	condition number	condition	formula
b197	Life insurance after apr 84-amnt	set24	1	lingtone in (1.2.4.5) and lingston—2	ours (duling)
0197	prmum  Endow policies after apr 84-amnt	set24	1	linstype in (1,3,4,5) and linsstyr =2	sum(dvlins)
b198	prmum	set12	1	linstype = $2$ and linsstyr = $2$	sum(dvlins)
b199	Endow policies before apr 84- amnt prmum	set12	1	linstype = 2 and linsstyr = 1	sum(dvlins)
0177	Mortgage intrst\\prncpl - last	50112	1	inistype = 2 and inisstyr = 1	Sum(GVIIIIS)
b200	payment	set12	1	morttype in (2,5)	dvmort
b200	Mortgage intrst\\prncpl - last payment	set12	5	morttype in $(2,5)$ and strinc = 1 and striprem > 9997 and stritype = 1 and strival >	avg(b200) - sum(strival * 0.2/5200)
0200	Mortgage intrst\\prncpl - last	50112		moraype in (2,5) and same = 1 and surprem > >>>  and surrype = 1 and surran	avg(b200) - sum(strival * 0.2/5200) - sum(persival *
b200	payment	set12	6	morttype in (2,5) and strinc = 1 and striprem $>$ 9997 and stritype = 2 and strival $>$ 0 and persival $>$ 0	
b200	Mortgage intrst\\prncpl - last payment	set12	7		avg(b200) - sum(persival * 0.4 * 3.5/5200) - sum(persival
6200	Mortgage intrst\\prncpl - last	Set12	/	morttype in $(2,5)$ and strinc = 1 and striprem > 9997 and stritype = 2 and strival = 0 and persival > 0	avg(b200) - sum(strival * 0.2/5200) - sum(strival *
b200	payment	set12	8	morttype in (2,5) and strinc = 1 and striprem $>$ 9997 and stritype = 2 and strival $>$ 0 and persival = 0	(0.2/5200)/3.5)
	Mortgage endow pol after 1983				
b201	amount premium	set12	1	menpol = 1 and $menstyr = 2$	sum(dvmen)
	Mortgage endow pol pre 1984				
b202	amount premium	set12	1	menpol = 1 and menstyr = 1	sum(dvmen)
b203	Mixed mortgage - last payment	set12	1	morttype = 5	dvmort
b203	Mixed mortgage - last payment	set12	2	morttype = 5 and $whoinc = 2$	b203 - dvwho
				morttype = 5 and strinc = 1 and striprem > 9997 and stritype = 1 and strival	
b203	Mixed mortgage - last payment	set12	5	>0	avg(b203) - sum(strival * 0.2/5200)
				morttype = 5 and strinc = 1 and striprem $>$ 9997 and stritype = 2 and strival $>$ 0 and	avg(b203) - sum(strival * 0.2/5200) - sum(persival *
b203	Mixed mortgage - last payment	set12	6	*	0.4/5200)
b203	Mixed mortgage - last payment	set12	7	morttype = 5 and strinc = 1 and striprem > 9997 and stritype = 2 and strival = 0 and persival > 0	avg(b203) - sum(persival * 0.4 * 3.5/5200) - sum(persival * 0.4/5200)
	Other insurance - total amount				
b206	premium	set24	1	minstype = 3	sum(dvmins)
	Mrtge protection pre apr 84-amnt				
b208	prmum	set12	1	incmstyr = 1	sum(dvmorp)
	Mrtge protection after apr 84-amnt				
b213	prmum	set12	1	incmstyr = 2	sum(dvmorp)
	Medical insurance - total amount				
b229	premium	set24	1	minstype in (1,2)	sum(dvmins)
b303	Pay-amount last time paid	set43	1		dvpay - dvmil

variable	Variable description	table name	condition number	condition	formula
b312	Bonus earnings - gross amount	set43	2	bobatax = 2	avg(b312) + sum((dvbon * 100)/78.0)
b330	Bonus earnings - net amount	set43	2	bobatax = 1	avg(b330) + sum(dvbon * 0.78)
b334	Money sent abroad - amount	set54	1		dvpri
b342	Statutory maternity pay-amount received	set46	1	bens2_numwks1 <= 6 and a254 = 2 and dvusgr >= 61 and (bens2_bncards = 2 or bens2_bncards1 = 2)	dvusgr * 0.9
b342	Statutory maternity pay-amount received	set46	2	bens2_numwks1 <= 6 and a254 = 1 and wkgross >= 61 and (bens2_bncards = 2 or bens2_bncards1 = 2)	wkgross * 0.9
b342	Statutory maternity pay-amount received	set46	3	bens2_numwks1 > 6 and bens2_numwks1 <= 18	60.2
b348	Pension(public\\private)last net payment	set47	2	penstype in (1,2) and ptinc = 1	avg(b348) - sum(dvtaxa)
b388	Statutory sick pay-imputed gross amount	set49	1	(bens1_bncard00 = 5 or bens1_bncard01 = 5 or bens1_bncard02= 5 or bens1_bncard03 = 5 or bens1_bncard04 = 5 or bens1_bncard05= 5) and ina254 = 1	wkgross
b388	Statutory sick pay-imputed gross amount	set49	2	(bens1_bncard00 = 5 or bens1_bncard01 = 5 or bens1_bncard02 = 5 or bens1_bncard03 = 5 or bens1_bncard04 = 5 or bens1_bncard05 = 5) and ina254 = 2	dvusgr
b388	Statutory sick pay-imputed gross amount	set49	4	b388 >= 61	60.2
b404	Disable wrking allw/Disable Pers Tax Cr.	set49	1		dvdptc2
d010101u	Rent - diary expenditure	setunanonh	1	dqualif in (0,3,4) and ditemcode = '010101' and personcopy < 100	sum(ditemamt/2)
d010102	Mortgage instalment payment	set900	1	dqualif in (0,3,4) and ditemcode = '010102' and personcopy < 100	sum(ditemamt/2)
d010103	Mortgage endowment policy	set900	1	dqualif in (0,3,4) and ditemcode = '010103' and personcopy < 100	sum(ditemamt/2)
d010104	Mortgage protection policy	set900	1	dqualif in (0,3,4) and ditemcode = '010104' and personcopy < 100	sum(ditemamt/2)
d010105u	Council tax and rates-diary expenditure	setunanonh	1	dqualif in (0,3,4) and ditemcode = '010105' and personcopy < 100	sum(ditemamt/2)
	Water rates - diary expenditure	setunanonh		dqualif in (0,3,4) and ditemcode = '010106' and personcopy < 100	sum(ditemamt/2)
d010107	Service charges	set900		dqualif in (0,3,4) and ditemcode = '010107' and personcopy < 100	sum(ditemamt/2)
d010108	Structural insurance	set900		dqualif in (0,3,4) and ditemcode = '010108' and personcopy < 100	sum(ditemamt/2)
d010109	Contents insurance	set900		dqualif in $(0,3,4)$ and ditemcode = $'010109'$ and personcopy $< 100$	sum(ditemamt/2)
d010110	Ground rent	set900	l	dqualif in (0,3,4) and ditemcode = '010110' and personcopy < 100	sum(ditemamt/2)
d010301	Central heating installation(contractor)	set900	1	dqualif in (0,3,4) and ditemcode = '010301' and personcopy < 100	sum(ditemamt/2)
d010302	Central heating maintenance (contractor)	set900	1	dqualif in (0,3,4) and ditemcode = '010302' and personcopy < 100	sum(ditemamt/2)

variable	Variable description	table name	condition number	condition	formula
•					
d010303	Capital improvements (contractor)	set900	1	dqualif in (0,3,4) and ditemcode = '010303' and personcopy < 100	sum(ditemamt/2)
	Repairs, decor, replacements				
d010304	(contractor)	set900	1	dqualif in (0,3,4) and ditemcode = '010304' and personcopy < 100	sum(ditemamt/2)
d020101	Gas account payment	set900	1	dqualif in (0,3,4) and ditemcode = '020101' and personcopy < 100	sum(ditemamt/2)
d020102	Gas board budgeting payment	set900	1	dqualif in (0,3,4) and ditemcode = '020102' and personcopy < 100	sum(ditemamt/2)
d020201	Electricity account payment	set900	1	dqualif in (0,3,4) and ditemcode = '020201' and personcopy < 100	sum(ditemamt/2)
	Electricity board budgeting				
d020202	payment	set900	1	dqualif in (0,3,4) and ditemcode = '020202' and personcopy < 100	sum(ditemamt/2)
d020302	Central heating oil	set900	1	dqualif in (0,3,4) and ditemcode = '020302' and personcopy < 100	sum(ditemamt/2)
d020304	Bottled gas	set900	1	dqualif in (0,3,4) and ditemcode = '020304' and personcopy < 100	sum(ditemamt/2)
d031302	State school meals	set900	1	dqualif in (0,3,4) and ditemcode = '031302' and personcopy < 100	sum(ditemamt/2)
1070101		.000		1 101 (0.0 t) 1 11 10701011 1	(1)
d070101	Furniture inc beds and mattresses	set900		dqualif in $(0,3,4)$ and ditemcode = $ 070101 $ and personcopy $< 100$	sum(ditemamt/2)
d070102	Soft floor coverings Telephone account	set900		dqualif in $(0,3,4)$ and ditemcode = '070102' and personcopy < 100	sum(ditemamt/2) sum(ditemamt/2)
d080203	Telephone account	set900	1	dqualif in (0,3,4) and ditemcode = '080203' and personcopy < 100	sum(ditemamt/2)
d080206	Mobile phone account payments	set900	1	dqualif in (0,3,4) and ditemcode = '080206' and personcopy < 100	sum(ditemamt/2)
d080200 d080402	Bank service charges	set900	1	dqualif in (0,3,4) and ditemcode = '080402' and personcopy < 100'	sum(ditemant/2)
0000402	Conveyancing, est agents,	861900	1	uquani iii (0,3,4) and ditenicode = 000402 and personcopy < 100	sum(uitemami/2)
d080404	surveyors fees	set900	1	dqualif in (0,3,4) and ditemcode = '080404' and personcopy < 100	sum(ditemamt/2)
4080404	surveyors rees	SCLOO	1	uquam in (0,3,4) and ditentedee = 000404 and personcopy < 100	Sum(dicinamo 2)
d100101	New car or van outright purchase	set900	1	dqualif in (0,3,4) and ditemcode = '100101' and personcopy < 100	sum(ditemamt/2)
<u>u100101</u>	The wear of variouright parenase	3000	-	aquani in (0,3,4) and diemeode = 100101 and personeopy \ 100	Sum(diterrating 2)
d100102	Second hand car outright purchase	set900	1	dqualif in (0,3,4) and ditemcode = '100102' and personcopy < 100	sum(ditemamt/2)
d100103	Car leasing payments	set900		denote by denot	sum(ditemamt/2)
	New, second hand motor cycle			(,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
d100104	purchase	set900	1	dqualif in (0,3,4) and ditemcode = '100104' and personcopy < 100	sum(ditemamt/2)
d100201	Car or van repairs and servicing	set900	1	dqualif in (0,3,4) and ditemcode = '100201' and personcopy < 100	sum(ditemamt/2)
d100204	Motor cycle repairs and servicing.	set900	1	dqualif in (0,3,4) and ditemcode = '100204' and personcopy < 100	sum(ditemamt/2)
d100401	Vehicle insurance	set900	1	dqualif in (0,3,4) and ditemcode = '100401' and personcopy < 100	sum(ditemamt/2)
d100402	Vehicle tax	set900	1	dqualif in (0,3,4) and ditemcode = '100402' and personcopy < 100	sum(ditemamt/2)
d110201	Railway and tube season tickets	set900	1	dqualif in (0,3,4) and ditemcode = '110201' and personcopy < 100	sum(ditemamt/2)
d110203	Bus and coach season tickets	set900		dqualif in (0,3,4) and ditemcode = '110203' and personcopy < 100	sum(ditemamt/2)
d110205	Combined fare season tickets	set900	1	dqualif in (0,3,4) and ditemcode = '110205' and personcopy < 100	sum(ditemamt/2)

			condition		
variable	Variable description	table name	number	condition	formula
d110210	Diary expenditure on school travel	set14	1	pdrcode = '110210'	sum(pdcodval)
d110211	Water travel season ticket	set900		dqualif in (0,3,4) and ditemcode = '110211' and personcopy < 100	sum(ditemamt/2)
d130201	TV licences	set900	1	dqualif in (0,3,4) and ditemcode = '130201' and personcopy < 100	sum(ditemamt/2)
	TV, video recorder, satellite TV				
d130203	rental	set900	1	dqualif in (0,3,4) and ditemcode = '130203' and personcopy < 100	sum(ditemamt/2)
	Fees, maintenance: educational				
d130301	courses	set900	1	dqualif in (0,3,4) and ditemcode = '130301' and personcopy < 100	sum(ditemamt/2)
	Fees, maintenance: non-household				
d130302	member	set900	1	dqualif in (0,3,4) and ditemcode = '130302' and personcopy < 100	sum(ditemamt/2)
d130303	Fees for leisure classes	set900	1	dqualif in (0,3,4) and ditemcode = '130303' and personcopy < 100	sum(ditemamt/2)
	Holidays in UK: accomm (exc				
d130401	timeshares)	set900	1	dqualif in (0,3,4) and ditemcode = '130401' and personcopy < 100	sum(ditemamt/2)
	Holidays abroad: accomm (exc				
d130402	timeshares)	set900	1	dqualif in (0,3,4) and ditemcode = '130402' and personcopy < 100	sum(ditemamt/2)
	Timeshares, holiday homes				
d130403	overseas	set900	1	dqualif in (0,3,4) and ditemcode = '130403' and personcopy < 100	sum(ditemamt/2)
d140102	Private personal pension	set900	1	dqualif in (0,3,4) and ditemcode = '140102' and personcopy < 100	sum(ditemamt/2)
	Money set aside for payment of				
d140104	bills	set900	1	dqualif in (0,3,4) and ditemcode = '140104' and personcopy < 100	sum(ditemamt/2)
	Life,death insurance,non-house				
d140201	endowment	set900		dqualif in (0,3,4) and ditemcode = '140201' and personcopy < 100	sum(ditemamt/2)
d140202	Private medical insurance	set900	1	dqualif in (0,3,4) and ditemcode = '140202' and personcopy < 100	sum(ditemamt/2)
	Accident,sick,redundancy,other				
d140203	insurance	set900	1	dqualif in (0,3,4) and ditemcode = '140203' and personcopy < 100	sum(ditemamt/2)
d140304	Money sent abroad	set900	1	dqualif in (0,3,4) and ditemcode = '140304' and personcopy < 100	sum(ditemamt/2)
	Maintenance or separation	0.00			(1)
d140305	allowance	set900	l	dqualif in (0,3,4) and ditemcode = '140305' and personcopy < 100	sum(ditemamt/2)
11.40206	Money given to other household	.000		1 1121 (0.0 4) 1 11 11 11 11 11 11 11 11 11 11 11 11	(1)
d140306	spenders	set900	1	dqualif in (0,3,4) and ditemcode = '140306' and personcopy < 100	sum(ditemamt/2)
11.40200	D 1 ( 131 13	12.41		1 1'6' (0.2) 1.1'. 1 11402001 1 100	(1)
d140308t	Pocket money to child no diary	set24t	1	dqualif in (0,3) and ditemcode = '140308' and personcopy < 100	sum(ditemamt/2)
11.40200:	Money given to child for specific	+0.44	1	J1:f: (0.2) J 11402001 - 1 100	aum/ditamamt/2)
d140309t	purpos Cash gifts to children	set24t	1	dqualif in (0,3) and ditemcode = '140309' and personcopy < 100 dqualif in (0,3) and ditemcode = '140310' and personcopy < 100	sum(ditemant/2)
	ĕ	set24t			sum(ditemant/2)
d140401	Credit card account payment	set900	1	dqualif in (0,3,4) and ditemcode = '140401' and personcopy < 100	sum(ditemant/2)
d140402	Credit card interest payment Credit card ann standing charge	set900	1	dqualif in (0,3,4) and ditemcode = '140402' and personcopy < 100	sum(ditemamt/2)
4140402	0 0	aat000	1	develif in (0.2.4) and ditamaged = 11404021 and assessment 100	sum(ditemamt/2)
d140403	payment	set900	1	dqualif in (0,3,4) and ditemcode = '140403' and personcopy < 100	
d140404	Loan instalment payment	set900	1	dqualif in (0,3,4) and ditemcode = '140404' and personcopy < 100	sum(ditemamt/2)

			condition		
variable	Variable description	table name		condition	formula
	Hire purchase instalment payment	set900	1	dqualif in (0,3,4) and ditemcode = '140405' and personcopy < 100	sum(ditemamt/2)
d140406	Club instalment payment	set900		dqualif in (0,3,4) and ditemcode = '140406' and personcopy < 100	sum(ditemamt/2)
d140501	Income tax payment	set900	1	dqualif in (0,3,4) and ditemcode = '140501' and personcopy < 100	sum(ditemamt/2)
d140502	National insurance contribution	set900	1	dqualif in (0,3,4) and ditemcode = '140502' and personcopy < 100	sum(ditemamt/2)
d140601	Household expenditure nos	set900	1	dqualif in (0,3,4) and ditemcode = '140601' and personcopy < 100	sum(ditemamt/2)
fs113	Air & other travel & transport	set26a	1		d110207t + d110208t + d110206t + b216 + d110301t + d110303t + d110302t + b248 + d110209t + b219 + d110210t
	Number of persons not				
g021	economically activ	set2	1	a206 in (6,7,0)	count(a206)
p003	Gross wage,salary last time paid (main)	set43	2		p003 + dvuni + dvspt + dvrem + dvavc+ dvfrie + dvins + wevchty
p003	Gross wage,salary last time paid (main)	set43	3		avg(p003) + sum (dvdedo)
1	Gr wage,salary lst time pd(main-				
p005	13 wks)	set43	1	a208 < 14	p003
p007	Normal gross wage,salary	set43	1		p006 + b305 + b306 + b318 + b371 + b372
					p007 + dvuni + dvspt + dvrem + dvavc+ dvfrie + dvins +
p007	Normal gross wage,salary	set43	2		wevchty
	Normal gross wage,salary(13 wk				
p008	rl)	set43	1	a208 < 14	p007
	Soc-sec bnfts concurrent with				
p028	earnings	set51	1	a250 = 1	p024 + p025 + p026
	Soc-sec bnfts included in income				p013 + p016 + p028 + b337 + b338 + b339 + b340 + b343 + b394 + b421 + b403 + b404 + b405 + b418 +
p031	calcs	set51	1		b406 + p085 + b335
p048	Income from investments	set46	1		b333 + b360 + b373 + b398 + b413 + b384 + b411 + ((b378 + b409 + b414) * 1.25)
p050	Income from other sources	set46	1		b316 + b351a + b381 + b359 + b396 + b364
p050	Income from other sources	set46	2	p005 = 0	p050 + b501
p066	Paye tax refund, 13 week rule	set44	1	a208 < 14	b304
	Income tax,payments less rfnds(13				p065 + b347 + b349 + b387 - (p066 + b390) + ((b378 +
p067	wk rl)	set44	1		b409 + b414) * 0.25) + (b312 - b330)
p068	Tax dctd from pensions,invests,annuities	set46	1		b347 + b349 + ((b378 + b409 + b414) * 0.25)
p079	Income tax payments less refunds	set44	1		p077 + b347 + b349 + b387 - (p078 + b390) + ((b378 + b409 + b414) * 0.25) + p073

variable	Variable description	table name	condition number	condition	formula
variable	variable description	tuore nume	number	Condition	Tormeta
	Int & tax on bank,nat				
p081	savings,TESSA,ISAs	set46	1		b333 + b373 + b398 + b411 + b413 + (b414 * 1.25)
				itemdea_pabl0001 = '100105' and (startdat - (date('15/' +	
100				varchar(loandea_loandatm) + '/' + varchar(loandea_loandaty))) <= '3 months') and	(1) 1 1 (12.0)
p108	New cars - purchase price	set54	1	itemdea_loannum = loandea_loannum agree1_pabh0011 = '100105' and startdat - (date('15/' + varchar(hpm) + '/' +	sum(itemdea_lcashp / 13.0)
p108	New cars - purchase price	set54	2	varchar(hpy)))<= '3 months' varchar(hpy))	avg(p108) + sum(cashhp / 13.0)
p100	riew cars parenase price	5005		agree2_pabh0001 = '100105' and startdat - (date('15/' + varchar(hpdatm) + '/' +	uvg(p100) + sum(cusimp + 13.0)
p108	New cars - purchase price	set54	3	varchar(hpdaty))) <= '3 months'	avg(p108) + sum(agree2_cashprce / 13.0)
				itemdea_pabl0001 = '100106' and (startdat - (date('15/' +	
100				varchar(loandea_loandatm) + '/' + varchar(loandea_loandaty))) <= '3 months') and	4
p109	Second-hand cars -purchase price	set54	1	itemdea_loannum = loandea_loannum	sum(itemdea_lcashp / 13.0)
p109	Second-hand cars -purchase price	set54	2	agree1_pabh0011 = '100106' and startdat - (date('15/' + varchar(hpm) + '/' + varchar(hpy))) <= '3 months'	avg(p109) + sum(cashhp / 13.0)
p109	Second-hand cars -purchase price	86134		agree2_pabh0001 = '100106' and startdat - (date('15/' + varchar(hpdatm) + '/' +	avg(p109) + sum(casimp / 13.0)
p109	Second-hand cars -purchase price	set54	3		avg(p109) + sum(agree2_CashPrce / 13.0)
p103	percona mana cars paremase price	5005		agree2_pabh0001 = '100107' and startdat - (date('15/' + varchar(hpdatm) + '/' +	a 'g(p10)'   5am(agree2_casm100 / 13.0)
p110	Motor cycles cars - purchase price	set54	3		avg(p110) + sum(agree2_cashprce/ 13.0)
				itemdea_pabl0001 = '100107' and (startdat - (date('15/' + varchar(loandea_loandatm) + '/'	
				+ varchar(loandea_loandaty))) <= '3 months') and itemdea_loannum =	
p110	Motor cycles cars - purchase price	set54	1	loandea_loannum	sum(itemdea_lcashp / 13.0)
- 110	M-4	451	2	agree1_pabh0011 = '100107' and startdat - (date('15/' + varchar(hpm) + '/' + varchar(hpy))) <= '3 months'	(-110) +(
p110	Motor cycles cars - purchase price Miscellaneous - personal	set54	2	varcnar(npy))) <= 3 months	avg(p110) + sum(cashhp / 13.0)
p126	expenditure	set54	1	pdrcode in (040801,0907011,403081,403091,40310)	sum(pdcodval)
p120	Miscellaneous - personal	5005	1	pareode in (040001,0707011,403001,403071,40310)	sum(pacoavar)
p126t	expenditure	set54t	1	pdrcode in (070801,090701)	sum(pdcodval)
•	Miscellaneous - personal			,	*
p126t	expenditure	set54t	2	pdrcodec in (070801,090701,140406,140601)	avg(p126t) + sum(pdcdvalc)
	Miscellaneous - personal				
p126t	expenditure	set54t	3	ditemcode in (140308, 140309, 140310) and dqualif in (0,3) and personcopy < 3	avg(p126t) + sum(ditemamt/2)
10.1	Fares etc (RPI)-personal	.=.			( )
p134	expenditure	set54	1	pdrcode like '11%' and pdrcode not in (110201,110203,110205,110211)	sum(pdcodval)
n124a	Fares etc (RPI)-personal expenditure	sot <b>5</b> 4 a	1	ndroodee like 110% and ndroodee not in (110201 110202 110205 110211)	cum(ndodyala)
p134c	expenditure	set54c	1	pdrcodec like '11%' and pdrcodec not in (110201,110203,110205,110211)	sum(pdcdvalc)
	Leisure services (RPI)- personal			pdrcode like '1301%' or pdrcode like '1305%' or pdrcode in (010607, 080303, 130202,	
p136	expend.	set54	1		sum(pdcodval)
	Soc-sec bnft(empl absnt w-o pay				
p175	2+wks)	set51	1	a200 = 1  and  208 > 2	p010 + p024 + p025 + p026 + p173

Description		1				
Soc-sec baffemplabant w-o pay   p175   2-wks   set14   1   a230 > 0   mod a240 = 1   p010 + p024 + p025 + p026 + p173 + b341   p184   Charisable gifts   set44   1   a230 > 0   weverby   weekpt   p184   Charisable gifts   set44   1   a230 > 0   weverby   weekpt   p184   Charisable gifts   set44   4   a2352 > 0 and poymatino publical = 140303   weights   samidvickido   p184   Charisable gifts   set44   4   a2552 > 0 and paymatino publical = 140303   weights   samidvickido   p184   charisable gifts   set44   4   a2552 > 0 and paymatino publical = 140303   weights   samidvickido   p281   p184   charisable gifts   set44   4   a2552 > 0 and pay2onatino publical = 140303   weights   samidvickido   p281   p184   charisable gifts   set44   a   a2552 > 0 and pay2onatino publical = 140303   weights   samidvickido   p100 + b100 + b130 + b150   p100 + b020 + b103 + b130 + b150   p100 + b020 + b101 + b130 + b150   p100 + b020 + b101 + b130 + b150   p100 + b100 + b10			. 11			c 1
p15   2 - wks   set5   2   200   1 and a208 > 2 and 240 = 1   p101 + p024 + p025 + p026 + p173 + b341     p184	variable	Variable description	table name	number	condition	formula
p15   2 - wks   set5   2   200   1 and a208 > 2 and 240 = 1   p101 + p024 + p025 + p026 + p173 + b341     p184		G 1 6/ 1 1 4		1		
Part	n175		aat 5 1	2	o200 = 1 and o208 > 2 and o240 = 1	n010 + n024 + n025 + n026 + n173 + h241
District   District	-	,				
Part   Charitable gifs   Set44   4   a2552 > 0 and pay2onum = 2 and pay2o pabded1 = '140303'   avg(p184) + sum(dvdeds)   b010 + b020 + b030 + b038 p + b050 p + b053 p + b056 p + b059 p + b05	1 -			_		
Net rentrates,coun tax,water,oth   bing   bing   set12   1     bing	<u> </u>					
P281p	p164		30144	4	a2332 > 0 and pay20ndin = 2 and pay20_pabded1 = 140303	
Net rentrates,coun tax,water,oth   bill	p281p		set12	1		
Disposable income   set3   1	F-0-F					b010 + b020 + b030 + b038u + b050 + b053u + b056u +
Disposable income   set3   1     sum(p177)	p281u		setunanonh	1		b060 + b110 + b130 + b150
Disposable income + income in p399   kind   set3   1   p365p = 0 and a099 = 1 and (p344 + p220p) < 1315.50   1   most households   set1   1   p365p = 0 and a099 = 1 and (p344 + p220p) < 1315.50   1   most households   set1   1   p365p = 0 and a099 = 1 and (p344 + p220p) < 1315.50   1   most households   set1   1   p365u = 0 and a099 = 1 and (p344 + p220p) < 1315.50   1   most households   set1   1   sum(p134) + avg(b216 + b217 + b218 + b248 + b219)   p526c   Fares and other travel costs   set14c   1   sum(p134) + avg(b216 + b217 + b218 + b248 + b219)   sum(p134) + avg(b216 + b217 + b218 + b248 + b219)   sum(p150) + avg(b160 + b162 + b164 + b181 + b194 + b265 + b480 + b481 + b485 + b483 + b48		Normal weekly disposable hhld				
P399   kind   set3   1   sum(p129)   midex households   set1   1   p365p = 0 and a099 = 1 and (p344 + p220p) < 1315.50   1	p389		set3	1		sum(p177)
P492p   index households		Disposable income + income in				
P492u   index households			set3	1		avg(p344 + b259 + b263 - (p388+p392)) - sum(p029)
Post   Fares and other travel costs   Set14   1   Sum(p134) + avg(b216 + b217 + b218 + b248 + b219)	p492p		set1	1		1
P526c   Fares and other travel costs   set14c   1   sum(p134c)	p492u	index households	setunanonh	1	p365u = 0 and $a099 = 1$ and $(p344 + p220u) < 1315.50$	1
P526c   Fares and other travel costs   set14c   1   sum(p134c)						
Post   Fares and other travel costs   Set   1   Sum(p134) + avg(b216 + b217 + b218 + b248 + b219)				_		
Sum(p136) + avg(b160 + b162 + b164 + b181 + b194 + b265 + b480 + b481 + b485 + b483 + b484 + b485 + b483 + b484 + b485 + b483 + b484 + b485 + b334h)   Possible	p526c	Fares and other travel costs	set14c	1		sum(p134c)
Sum(p136) + avg(b160 + b162 + b164 + b181 + b194 + b265 + b480 + b481 + b485 + b483 + b484 + b485 + b483 + b484 + b485 + b483 + b484 + b485 + b334h)   Possible	n526t	Fares and other travel costs	set1/lt	1		sum(n134t) + avg(h216 + h217 + h218 + h248 + h210)
Description	p320t	Tares and other traver costs	301171	1		
Description						* '
p530   Permanent second dwelling   set12   1   sum(p129) + avg(b101 + b102 + b106 + b108)     sum(p129) + avg(b101 + b102 + b106 + b108)     sum(p128) + avg(b101 + b103 + b105 + b106 + b196 + b197 + b198 + b199 + b200 + b201 + b202 + b203 + b2     p531   Other payments (neg acceptable)   set24   1   sum(p127) + avg(b101 + b103 + b105 + b106 + b130 + b200 + b203)     p402   p546   ONS fares and other travel costs   set24   1   sum(p127) + avg(b101 + b103 + b105 + b106 + b130 + b200 + b203)     p546   ONS fares and other travel costs   set24   1   sum(p134) + avg(b216 + b217 + b218 + b248 + b219)     p546   ONS fares and other travel costs   set24   1   sum(p134t) + avg(b216 + b217 + b218 + b248 + b219)     p547   ONS leisure services   set24   1   sum(p134t) + avg(b216 + b217 + b218 + b248 + b219)     p548   ONS leisure services   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p549   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p540   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p540   Miscellaneous   set24   1   sum(p126) + avg(b238 + b237)     p540   Miscellaneous   set24   1   sum(p127) + avg(b238 + b237)     p540   Miscellaneous   set24   1   sum(p127) + avg(b238 + b237)     p	p528	Leisure services	set20	1		
Sum(p128) + avg(b101 + b103 + b105 + b106 + b196 + b197 + b198 + b199 + b200 + b201 + b202 + b203 + b2 + b229 + p087 - b150)				1		sum(p129) + avg(b101 + b102 + b106 + b108)
Description	•	Ţ.				
Purchase of dwlng, imprvments & set24						b197 + b198 + b199 + b200 + b201 + b202 + b203 + b228
p532         mort pay         set24         1         sum(p127) + avg(b101 + b103 + b105 + b106 + b130 + b200 + b203)           p546         ONS fares and other travel costs         set24         1         sum(p134) + avg(b216 + b217 + b218 + b248 + b219)           p546c         ONS fares and other travel costs         set24c         1         sum(p134c)           p546t         ONS fares and other travel costs         set24c         1         sum(p134t) + avg(b216 + b217 + b218 + b248 + b219)           p548         ONS leisure services         set24c         1         p528c           p549         Miscellaneous         set24c         1         sum(p126) + avg(b238 + b237)           p549t         Miscellaneous         set24c         1         sum(p126t) + avg(b238 + b237)	p531		set24	1		+ b229 + p087 - b150)
p546         ONS fares and other travel costs         set24         1         sum(p134) + avg(b216 + b217 + b218 + b248 + b219)           p546c         ONS fares and other travel costs         set24c         1         sum(p134c)           p546t         ONS fares and other travel costs         set24t         1         sum(p134t) + avg(b216 + b217 + b218 + b248 + b219)           p548         ONS leisure services         set24t         1         p528           p549         Miscellaneous         set24t         1         sum(p126) + avg(b238 + b237)           p549t         Miscellaneous         set24t         1         sum(p126t) + avg(b238 + b237)		Purchase of dwlng, imprvments &				
p546c         ONS fares and other travel costs         set24c         1         sum(p134c)           p546t         ONS fares and other travel costs         set24t         1         sum(p134t) + avg(b216 + b217 + b218 + b248 + b219)           p548         ONS leisure services         set24         1         p528           p549         Miscellaneous         set24         1         sum(p126) + avg(b238 + b237)           p549t         Miscellaneous         set24t         1         sum(p126t) + avg(b238 + b237)	p532	mort pay	set24	1	sum(p127) + avg(b101 + b103 + b105 + b106 + b130 + b200 + b203)	
p546c         ONS fares and other travel costs         set24c         1         sum(p134c)           p546t         ONS fares and other travel costs         set24t         1         sum(p134t) + avg(b216 + b217 + b218 + b248 + b219)           p548         ONS leisure services         set24         1         p528           p549         Miscellaneous         set24         1         sum(p126) + avg(b238 + b237)           p549t         Miscellaneous         set24t         1         sum(p126t) + avg(b238 + b237)						
p546t         ONS fares and other travel costs         set24t         1         sum(p134t) + avg(b216 + b217 + b218 + b248 + b219)           p548         ONS leisure services         set24         1         p528           p549         Miscellaneous         set24         1         sum(p126) + avg(b238 + b237)           p549t         Miscellaneous         set24t         1         sum(p126t) + avg(b238 + b237)	p546	ONS fares and other travel costs	set24	1		sum(p134) + avg(b216 + b217 + b218 + b248 + b219)
p546t         ONS fares and other travel costs         set24t         1         sum(p134t) + avg(b216 + b217 + b218 + b248 + b219)           p548         ONS leisure services         set24         1         p528           p549         Miscellaneous         set24         1         sum(p126) + avg(b238 + b237)           p549t         Miscellaneous         set24t         1         sum(p126t) + avg(b238 + b237)			.24			( 124 )
p548         ONS leisure services         set24         1         p528           p549         Miscellaneous         set24         1         sum(p126) + avg(b238 + b237)           p549t         Miscellaneous         set24t         1         sum(p126t) + avg(b238 + b237)	p546c	UNS fares and other travel costs	set24c	1		sum(p134c)
p548         ONS leisure services         set24         1         p528           p549         Miscellaneous         set24         1         sum(p126) + avg(b238 + b237)           p549t         Miscellaneous         set24t         1         sum(p126t) + avg(b238 + b237)	n546t	ONS fares and other travel costs	set24t	1		sum(p134t) + avg(b216 + b217 + b218 + b248 + b219)
p549         Miscellaneous         set24         1         sum(p126) + avg(b238 + b237)           p549t         Miscellaneous         set24t         1         sum(p126t) + avg(b238 + b237)				1		
p549t Miscellaneous set24t 1 sum(p126t) + avg(b238 + b237)				1		r -
	<u> </u>			1		
	1					4 17 18 80 11 11 11
p551p values set12 1 b259 + b263 + p550p	p551p		set12	1		b259 + b263 + p550p

variable	Variable description	table name	condition number	condition	formula
variable	variable description	table name	namour	Condition	Tormon
	Total expenditure plus imputed				
p551tp	values	set24t	1		b259 + b263 + p550tp
	Total expenditure plus imputed				
p551tu	values	setunanonht	1		b259 + b263 + p550tu
	Total expenditure plus imputed				
p551u	values	setunanonh	1		b259 + b263 + p550u
					avg(d070801 + d090701 + d140308 + d140309 +
p553	Miscellaneous (national accounts)	set24	1		d140310) + sum(b396)
					avg(d070801t + d090701t + d140308t +d140309t +
p553t	Miscellaneous (national accounts)	set24t	1		d140310t + d140406c + d140601c) + sum(b396)
	Miscellaneous services - NAT				b237 + b238 + b265 + d130407 + d140302 + d140303 +
p559	ACCS	set24	1		d140307 + b334h
					b216 + b217 + b219 + d110206 +d110209 + d110301 +
p565	Other travel - NAT ACCS	set14	1		d110302 + d110303 + d110210
					d110206c + d110209c + d110301c + d110302c +
p565c	Other travel - nat accs	set14c	1		d110302c + d110303c + d110210c
p565t	Other travel - nat accs	set14t	1		p565 + p565c
					dvpay + dvtax1 + dvnino + wevchty + paymain_dvpen +
					dvuni + dvspt + dvrem + dvavc + dvfrie + dvins - (dvtax +
wkgross	Gross pay - amount	setwrk	1	paymain_payslip = 2	dvmil)
wkgross	Gross pay - amount	setwrk	3	paymain_payslip = 2	paymain_dvgros - dvmil
(207					

(207 rows)

end

Tue Oct 3 15:53:13 2000

Filename: /usr8/fes1/fes/newyear2000/derbands2000 Filename: derived banding changes in 2000-2001

This check produces a report that shows any variables that are common to both 1999 and 2000 BUT which have different banding names or have a banding entry in one year but not in the other.

This output is used to produce PART of SECTION 5, part 6						
	Banding changes year on year					

variable	description	table_name	banding	year_start	year_end
(0 rows)					

# **Derived CODING FRAMES IN 2000-2001**

Tue Oct 3 15:54:23 2000

filename: /fes1/users/fes1/fes/newyear2000/dercodingf2000

filename: derived CODING FRAMES in 2000/2001.xls

This check produces a report that shows any variables that are common to both 1999 and 2000 but which have an indicator to show that the **banding entry is a coding** 

**frame** in one year but not the other.

This output is used to produce PART of Section 5, Part 6. Previous year derived variables that have been amended for the current year

variable	description	table_name	year_start	year_end
d140402	Credit card interest payment	set24	1994	1999
d140402	Credit card interest payment	set900	2000	2999
2 rows				

Tue Oct 3 15:59:36 2000

file: /fes1/users/fes1/fes/newyear2000/dertype2000

file: derived data types 2000-2001.xls

The scripts in this file compare the metadata for 1999 derived variables with the 2000 derived variables.

This output is used to produce PART of **VOLUME 5**, **Part 6**Previous year derived variables that are DIFFERENT to the current year

This check produces a report that shows any variables that are common to both 1999 and 2000 but which have

#### DIFFERENT DATA TYPES

variable description table\_name year\_s year\_e dtype 0 rows

Tue Oct 3 15:58:29 2000

File: /fes1/users/fes1/fes/newyear2000/dertables2000

File: derived tables for 2000-2001.xls

The scripts in this file compare the metadata for 1999 derived tables with the 2000 derived tables.

The first check produces a report of tables that are in 1999 but not in 2000

table_name	year_start	year_end	level	extra_keys	derive	restricted	viewing
set26	1994	1999	h	n	d	n	
set53t	1995	1999	p	n	d	n	

2 rows

The second check produces a report of tables that are in 2000 but not in 1999

table_name	year_start	year_end	level	extra_keys	derive	restricted	viewing
0 rows							

The third check produces a report that shows any tables that are common to both 1999 and 2000 where they have changed person level.

year_start	year_end	level	extra_keys	derive	restricted	viewing
0 rows						

The FOURTH check produces a report that shows any tables that are common to both 1999 and 2000 but where they have changed from having extra index keys to not and vice versa.

table_name	year_start	year_end	level	extra_keys	derive	restricted	viewing
0 rows							